

# Working Paper No. 2016-01

# Optimal Policies to Promote Efficient Distributed Generation of Electricity

David P. Brown University of Alberta

# David E. M. Sappington University of Florida

January 2016

Copyright to papers in this working paper series rests with the authors and their assignees. Papers may be downloaded for personal use. Downloading of papers for any other activity may not be done without the written consent of the authors.

Short excerpts of these working papers may be quoted without explicit permission provided that full credit is given to the source.

The Department of Economics, the Institute for Public Economics, and the University of Alberta accept no responsibility for the accuracy or point of view represented in this work in progress.

# Optimal Policies to Promote Efficient Distributed Generation of Electricity

by

David P. Brown\* and David E. M. Sappington\*\*

#### Abstract

We analyze the design of policies to promote efficient distributed generation (DG) of electricity. The optimal policy varies with the set of instruments available to the regulator and with the prevailing DG production technology. DG capacity charges often play a valuable role in inducing optimal investment in DG capacity, allowing payments for DG production to induce the optimal production of electricity using non-intermittent DG technologies. Net metering can be optimal in certain settings, but often is not optimal, especially for non-intermittent DG technologies.

**JEL Categories**: D47, L50, L94, Q40 **Keywords**: electricity pricing, distributed generation, regulation

December 2015

\* Department of Economics, University of Alberta, Edmonton, Alberta T6G 2H4 Canada (dpbrown@ualberta.ca).

\*\* Department of Economics, University of Florida, Gainesville, FL 32611 USA USA (sapping@ufl.edu).

# 1 Introduction

The distributed generation of electricity is already pervasive in many countries and is expanding rapidly throughout the world.<sup>1</sup> Distributed generation (DG) is popular in part because it can limit the amount of capacity required at the primary production site, reduce electricity distribution costs (by moving generation sites closer to final consumers), and reduce generation externalities (e.g., carbon emissions).<sup>2</sup>

In addition to its many potential benefits, DG introduces at least two policy challenges that presently are the subject of heated debate.<sup>3</sup> First, depending upon the established retail prices for electricity and the terms of compensation for DG investment and production, electricity customers may undertake excessive or insufficient DG investment and production (Wiser et al., 2007; Borenstein, 2015). Second, as customers generate some or all of the electricity they consume, the centralized supplier of electricity ("the utility") experiences declines in revenue that typically exceed the associated avoided costs, given the large fixed infrastructure costs of the typical utility (Linvill et al. 2013; Lively and Cifuentes, 2014; Perez-Arriaga and Bharatkumar, 2014).

At least three changes have been proposed to help ensure the solvency of electric utilities while encouraging efficient DG investment and production. First, revised utility rate structures have been proposed. Rate structures that entail higher fixed charges for the right to purchase electricity from the utility and lower marginal charges for electricity actually purchased from the utility can better align the utility's revenues and costs in the presence of widespread DG (Brown and Faruqui, 2014; Costello, 2015).<sup>4</sup>

Second, various payments for DG production have been advocated, including alternatives

<sup>&</sup>lt;sup>1</sup>See (DNV GL, 2014), the World Alliance for Decentralized Energy (2014), and Solar Energy Industries Association (2015), for example. The distributed generation of electricity entails the "generation of electricity from sources that are near the point of consumption, as opposed to centralized generation sources such as large utility-owned power plants" (American Council for an Energy-Efficient Economy, 2015).

<sup>&</sup>lt;sup>2</sup>See Weissman and Johnson (2012), for example.

<sup>&</sup>lt;sup>3</sup>NCCETC (2015d) reviews recent and ongoing DG policy initiatives throughout the U.S.

<sup>&</sup>lt;sup>4</sup>Such rate restructuring can be viewed as a form of decoupling the utility's revenue from the amount of electricity it supplies. In addition to promoting efficient investment in DG capacity, decoupling can encourage a utility to promote energy conservation (Brennan, 2010a).

to common net metering policies. Under these policies, DG production causes a customer's electricity meter to run backward, so the consumer is effectively paid the prevailing retail price of electricity for each unit of electricity he produces on-site. Alternatives to such net metering include: (i) "value of production" feed-in tariffs, whereby a DG customer is compensated for the electricity he produces on-site at a rate that reflects the estimated net social value of the electricity (Farrell 2014a,b); and (ii) "avoided-cost" tariffs, whereby the unit payment for DG production reflects the associated reduction in the utility's production cost.

Third, additional charges on DG suppliers have been suggested. Several states have proposed that DG customers pay a fixed fee (independent of the level of the customer's installed DG capacity) to the utility. Proposed monthly fees vary from \$4.65 in Utah to \$25 in Maine.<sup>5</sup> DG charges that increase with a customer's generating capacity also have been implemented. For example, the Virginia Electric and Power Company imposes a monthly charge of \$4.19/kW on DG units with capacities between 10 and 20 kW (VSCC, 2011).<sup>6</sup> Additional charges on DG customers that increase with their maximum monthly consumption of electricity also have garnered attention (e.g., Hledik, 2014; Faruqui and Hledik, 2015).

Despite the widespread policy debate and the substantial variety of policies that have been proposed or implemented, formal analysis of the optimal design of DG compensation is limited.<sup>7</sup> This research is intended to help fill this void in the literature and to provide some

<sup>&</sup>lt;sup>5</sup>See Maine Public Utilities Commission (2013) and NCCETC (2015d).

<sup>&</sup>lt;sup>6</sup>The largest utility in Arizona presently imposes a fee of \$0.70/kW on solar DG capacity, and has suggested that this fee be raised to \$3.00/kW (NCCETC, 2015d). In 2014, the Wisconsin Public Service Commission approved a capacity charge of \$3.79/kW on wind and solar DG. See NCCETC (2015d) and filings in the Arizona Corporation Commission's Docket No. 13-0248 (http://edocket.azcc.gov/Docket/DocketDetailSearch?docketId=18039#docket-detail-container2) and in the Wisconsin Public Service Commission Docket No. 5-UR-107 (http://psc.wi.gov).

<sup>&</sup>lt;sup>7</sup>Industry experts have provided useful recommendations regarding various elements of DG compensation. Couture and Gagnon (2010), Kind (2013), and Raskin (2013), among others, review and discuss these recommendations. However, these recommendations typically do not reflect the explicit predictions of comprehensive, formal economic models. Darghouth et al. (2011, 2014) and Poullikkas (2013) simulate the effects of selected forms of DG compensation. Yamamoto (2012) models some considerations in the design of DG compensation policies, but abstracts from such elements as the full impact of DG investment on a consumer's energy costs. Brown and Sappington (2015a) examine whether common net metering policies are ever optimal in a setting where the regulator's retail pricing instruments are limited, DG capacity

guidance to policymakers on this important matter. We characterize the retail tariffs, payments for DG production, and charges for DG capacity investment that maximize consumer welfare while ensuring the solvency of the regulated utility. Our analysis allows for both intermittent and non-intermittent DG technologies. Solar panels constitute an intermittent technology in the sense that the amount of electricity produced by the installed capacity is largely exogenous, dictated primarily by environmental conditions (especially the prevailing level of sunshine). In contrast, resources such as fuel cells, gas turbines, and reciprocating engines (which employ natural gas as the primary fuel) constitute non-intermittent DG technologies because the amount of electricity produced by the installed capacity is readily controlled.<sup>8</sup>

We begin by analyzing a setting in which smart meters are deployed ubiquitously and the regulator can set real-time retail prices and DG payments. In this setting, the retail prices for electricity and the unit payments for DG production are both set equal to the utility's prevailing marginal cost of generating electricity under an optimal policy. These retail prices and DG payments ensure the efficient consumption and production of electricity, given installed generating capacities. Unit charges on installed DG capacity that reflect the associated increase (or decrease) in the utility's transmission and distribution costs then induce efficient DG capacity investment. The identical retail prices and DG payments imply that net metering as described above is optimal. Furthermore, the ability to set different retail prices or DG payments for different consumers or for different DG technologies would not enhance consumer welfare.

We also consider a setting where the regulator can only establish time-of-use retail prices and time-of-production DG payments. These prices and payments can vary across pre-

payments are not feasible, and only one (fully intermittent) DG production technology is available.

<sup>&</sup>lt;sup>8</sup>Solar panels account for the majority of DG capacity in most U.S. states, due in part to the rapid decline in the cost of solar panels in recent years (Barbose et al., 2014). However, natural gas-based DG configured in combined heat and power (CHP) mode account for the majority of DG capacity in Connecticut and New York (DNV GL, 2014). CHP units can be of particular value as a reliable alternative source of electricity when primary sources fail (U.S. Department of Energy, 2013).

specified time periods each day, but cannot vary within a time period. To induce desired levels of electricity consumption in this setting, the retail prices are set so that a weighted average of expected deviations of price from the utility's marginal cost of production is zero. In Ramsey-like fashion (Ramsey, 1927; Baumol and Bradford, 1970), the weights on deviations between price and cost reflect corresponding price sensitivities of the demand for electricity. To induce desired levels of electricity production from non-intermittent DG technologies, payments for such production are set equal to the utility's expected marginal cost of production during the relevant time period. To induce desired DG capacity investment, DG capacity charges are set to equate the consumer's marginal expected return from increasing DG capacity with the corresponding net reduction in the utility's production cost. This net reduction is the difference between the marginal reduction in the utility's transmission and distribution costs as DG capacity expands.

Because payments for DG production do not affect the amount of electricity produced with a given level of intermittent DG capacity, a regulator who can impose DG capacity charges has considerable flexibility in setting payments for electricity produced with the intermittent DG technology. In particular, the regulator can equate these payments either with the corresponding payments for electricity produced using the non-intermittent DG technology or with the corresponding retail rates for electricity. Consequently, neither a requirement to set the same DG payments for both technologies nor a requirement to implement the common net metering policy for the intermittent technology would be constraining in isolation. However, the imposition of both requirements simultaneously would reduce the level of welfare that can be secured for consumers.

The presence of social losses from externalities resulting from electricity production modify the foregoing findings in an intuitive fashion. In particular, retail electricity prices are optimally increased to discourage the purchase of electricity from the utility as the social losses from externalities associated with electricity production by the utility increase. In addition, DG payments for electricity produced using the non-intermittent technology increase as marginal social losses from externalities associated with this production decline relative to the marginal social losses associated with electricity production by the utility. Thus, more generous payments for electricity produced by "cleaner" DG technologies are optimal, *ceteris paribus*.

Overall, our findings imply that there is no single DG compensation policy that is optimal in all settings. The optimal policy varies with the instruments available to the regulator and with the relevant DG production technologies. Our findings also stress the important role of DG capacity charges. When these charges are feasible, the regulator can set them to induce efficient DG capacity investment while employing payments for DG production to induce efficient levels of electricity production, and net metering is optimal in certain settings. If the regulator is unable to impose capacity charges, then payments for DG production must serve to induce desired levels of both DG production and DG capacity investment. This dual role of payments for DG production complicates their design, limits the settings in which net metering is optimal, raises industry production costs, and reduces the level of welfare that can be secured for consumers.

We develop and explain these findings as follows. Section 2 presents the key features of our formal model. Section 3 characterizes the optimal policy in the setting where the regulator can set real-time retail prices and DG payments. Section 4 examines the changes to the optimal policy that arise in the presence of social losses from externalities. Section 5 identifies the key features of the optimal policy when the regulator can set time-of-use retail prices and time-of-production DG payments. Section 6 describes the optimal policy when the regulator's instruments are more limited, i.e., when retail rates cannot vary across demand periods or across customers, and when DG capacity charges are not feasible. Section 7 employs numerical solutions to illustrate how the optimal regulatory policy and industry outcomes change when the regulator is able to set DG capacity charges. Section 8 reviews the policy implications of our findings and suggests directions for further research.

## 2 Model Elements

A vertically-integrated producer (VIP) generates electricity, supplies it to consumers, and operates the sole transmission and distribution network.<sup>9</sup> There are two consumers, Dand N, who purchase electricity from the VIP. Consumer D also can undertake distributed generation (DG) of electricity, whereas consumer N cannot do so.<sup>10</sup>

There are two periods of demand for electricity, denoted low (L) and high (H). Consumer  $j \in \{D, N\}$  derives value  $V_t^j(x, \theta_s)$  from x units of electricity in state  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$  in period  $t \in \{L, H\}$ . This value is a strictly increasing, strictly concave function of x. The state variable  $\theta_s$  can be viewed as a measure of the amount of sunshine that prevails at a specified time in the relevant period. Therefore, in hot climates for example, higher realizations of  $\theta_s$  often will be associated with higher total and marginal valuations of electricity (to power cooling units).<sup>11</sup> The distribution function for the state variable in period t is  $F_t(\cdot)$ . The corresponding density function is  $f_t(\cdot)$ .

We will consider settings where the regulator can, and other settings where she cannot, set retail prices for electricity and payments for electricity produced via DG that vary with the realized state. In settings where the regulator can do so,  $r_{jt}(\theta_s)$  will denote the retail price that consumer j must pay to the VIP for each unit of electricity he consumes in state  $\theta_s$  in period t.<sup>12</sup> Consumer  $j \in \{D, N\}$  also must pay the fixed charge  $R_j$  for the right to purchase electricity from the VIP.<sup>13</sup>

Consumer j's demand for electricity in state  $\theta_s$  in period t is  $X_t^j(r_{jt}(\theta_s), \mathbf{r}_{\mathbf{jt}'}(\theta_{\mathbf{t}'}), \theta_s)$ , where  $\mathbf{r}_{\mathbf{jt}'}(\theta_{\mathbf{t}'})$  is the vector of retail prices that consumer j might potentially face in period

<sup>12</sup>In settings where the regulator cannot set state-specific prices,  $r_{jt}(\theta_s) = r_{jt}(\theta_{s'})$  for all  $\theta_s, \theta_{s'} \in [\underline{\theta}_t, \overline{\theta}_t]$ .

<sup>&</sup>lt;sup>9</sup>The setting we analyze also can be viewed as one in which the electricity supplier secures electricity via market transactions in a setting where a system operator dispatches supply in order of increasing cost.

<sup>&</sup>lt;sup>10</sup>We assume there are only two consumers for expositional ease. Our qualitative conclusions are unchanged if there are multiple D consumers and multiple N consumers.

<sup>&</sup>lt;sup>11</sup>Formally, in hot climates,  $\frac{\partial V_t^j(x,\theta_s)}{\partial \theta_s} \ge 0$  and  $\frac{\partial^2 V_t^j(x,\theta_s)}{\partial \theta_s \partial x} \ge 0$  for all  $x \ge 0$  and  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ , for  $j \in \{D, N\}$ . These inequalities need not hold more generally. The findings reported below hold even if these inequalities do not hold.

<sup>&</sup>lt;sup>13</sup>This formulation admits different retail prices for different consumers. The ensuing discussion also will consider settings in which such price discrimination is not feasible.

 $t' \neq t \ (t, t' \in \{L, H\})$ , depending on the realization of the state variable.<sup>14</sup> Aggregate consumer demand in state  $\theta_s$  in period t is  $X_t(\cdot, \theta_s) = X_t^D(\cdot, \theta_s) + X_t^N(\cdot, \theta_s)$ .

In addition to purchasing electricity from the VIP, consumer D can produce electricity himself and sell any excess of production over consumption to the VIP. Consumer D can produce electricity using an intermittent technology and/or a non-intermittent DG technology. The key difference between the technologies concerns the consumer's control over electricity production (dispatch) after installing generating capacity. For simplicity, we assume consumer D has no residual control over electricity produced using the intermittent DG technology (e.g., solar panels). Once he installs  $K_{Di}$  units of capacity of this technology, consumer D automatically produces  $Q_t^i(K_{Di}, \theta_s) = \mu_t \theta_s K_{Di}$  units of electricity in state  $\theta_s$ in period t at no incremental cost, where  $\mu_t \geq 0$  is an efficiency parameter.<sup>15</sup> Thus,  $K_{Di}$ might be viewed as the number of solar panels of a fixed size that consumer D installs, and each panel produces more electricity as the amount of sunshine increases at a rate that the consumer cannot control.<sup>16</sup>

In contrast, the consumer can control the amount of electricity generated by the nonintermittent DG technology he installs (e.g., a combined heat and power unit powered by natural gas). Increased investment in this technology  $(K_{Dn})$  reduces consumer D's variable and marginal cost of producing electricity in each demand period. Formally,  $\frac{\partial C_t^D(Q_t^n, K_{Dn})}{\partial K_{Dn}} < 0$ and  $\frac{\partial^2 C_t^P(\cdot)}{\partial K_{Dn} \partial Q_t^n} < 0$  for all  $Q_t^n > 0$ , where  $C_t^D(Q_t^n, K_{Dn})$  is consumer D's variable cost of producing  $Q_t^n$  units of electricity in period  $t \in \{L, H\}$  with the non-intermittent technology, given  $K_{Dn}$  units of capacity of the technology. This cost function is an increasing, convex function of  $Q_t^n$ .<sup>17</sup> Knowing this cost function and the prevailing payments for electricity pro-

<sup>17</sup>Formally, 
$$\frac{\partial C_t^D(Q_t^n, K_{Dn})}{\partial Q_t^n} > 0$$
 and  $\frac{\partial^2 C_t^D(Q_t^n, K_{Dn})}{\partial (Q_t^n)^2} > 0$  for all  $Q_t^n \ge 0$ 

 $<sup>{}^{14}</sup>X_t^j(r_{jt}(\theta_s), \mathbf{r_{jt'}}(\theta_{t'}), \theta_s)$  is a strictly decreasing function of  $r_{jt}(\theta_s)$  and a non-decreasing function of each element of  $\mathbf{r_{jt'}}(\theta_{t'})$ . Here and throughout the ensuing analysis, we abstract from income effects in assuming that the fixed charge does not affect a consumer's demand for electricity.

<sup>&</sup>lt;sup>15</sup>The precise placement of solar panels and the surrounding foliage or the adjacent structures at a specific location can cause electricity production to vary at different times of the day, holding constant the amount of sunshine that prevails in a geographic region.

<sup>&</sup>lt;sup>16</sup>The presumed linear relationship between output and capacity is adopted for expositional simplicity and does not affect the qualitative conclusions drawn below.

duced using the non-intermittent capacity he has installed, consumer D chooses to produce the profit-maximizing level of electricity in each state.

Consumer D's cost of installing  $K_{Di}$  units of capacity of the intermittent DG technology and  $K_{Dn}$  units of capacity of the non-intermittent DG technology is  $C_D^K(K_{Di}, K_{Dn})$ , which is a strictly increasing function of each of its arguments. For analytic ease, we focus on settings where consumer D installs strictly positive levels of capacity of both DG technologies.<sup>18</sup>

The regulator specifies the compensation that the VIP must deliver to consumer D for the DG capacity he installs.  $k_{0y}$  is the fixed payment the VIP must make to consumer Dwhen he installs any strictly positive amount of capacity in DG technology  $y \in \{i, n\}$ .  $k_y$ is the additional payment the VIP must make to consumer D for each unit of capacity of technology y he installs. These payments can be negative, in which case the VIP is authorized to impose capacity charges on consumer D.

The regulator also specifies the compensation the VIP must deliver to consumer D for the electricity he produces.  $w_{yt}(\theta_s)$  is the payment the VIP must make to consumer D for each unit of electricity he produces using technology  $y \in \{i, n\}$  in state  $\theta_s$  in period t.<sup>19</sup>

The VIP's variable cost of generating  $Q_t^v$  units of electricity in period t when it has  $K_G$  units of generating capacity is  $C_t^G(Q_t^v, K_G)$ . Increased generating capacity reduces at a diminishing rate the VIP's variable and marginal cost of generating electricity.<sup>20</sup> The VIP's cost of installing  $K_G$  units of generating capacity is  $C^K(K_G)$ , which is an increasing, convex function.<sup>21</sup>

<sup>20</sup>Formally,  $\frac{\partial C_t^G(Q_t^v, K_G)}{\partial K_G} < 0$ ,  $\frac{\partial^2 C_t^G(Q_t^v, K_G)}{\partial Q_t^v \partial K_G} < 0$ , and  $\frac{\partial^3 C_t^G(Q_t^v, K_G)}{\partial Q_t^v \partial^2 K_G} > 0$  for all  $Q_t^v > 0$ .

<sup>&</sup>lt;sup>18</sup>Consumer *D* will optimally do so in equilibrium if, for instance,  $\underset{K_{Dy} \to 0}{limit} \frac{\partial C_D^K(\cdot)}{\partial K_{Dy}} = 0$  for  $y \in \{i, n\}$  and  $\underset{K_{Dn} \to 0}{limit} \left| \frac{\partial C_t^D(\cdot)}{\partial K_{Dn}} \right| = \infty$  for  $t \in \{L, H\}$ .

<sup>&</sup>lt;sup>19</sup>The VIP must make this payment to consumer D regardless of whether consumer D consumes the unit of electricity in question. Recall that consumer D must pay the prevailing retail charge for each unit of electricity he consumes, regardless of the source of the electricity. Thus, the common net metering policy prevails if  $w_{yt}(\theta_s) = r_{jt}(\theta_s)$  in each period t and state  $\theta_s$ , for both technologies ( $y \in \{i, n\}$ ) and both consumers ( $j \in (D, N\}$ ).

<sup>&</sup>lt;sup>21</sup>Formally,  $C^{K'}(K_G) > 0$  and  $C^{K''}(K_G) > 0$ . We also assume a strictly positive level of generating capacity is optimal. This will be the case if, for example,  $\lim_{K_G \to 0} \left| \frac{\partial C^G_t(Q^v_t, K_G)}{\partial K_G} \right| = \infty$  and  $\lim_{K_G \to 0} C^{K'}(K^G) = 0$ .

The VIP also incurs transmission and distribution (T&D) costs  $T(K_G, K_{Di}, K_{Dn})$  to support centralized and distributed generating capacities  $K_G$ ,  $K_{Di}$ , and  $K_{Dn}$ .<sup>22</sup> For simplicity, we abstract from variable T&D costs associated with line losses that might arise even when adequate infrastructure is implemented to fully accommodate electricity produced by the installed generating capacity.<sup>23</sup>

The regulator chooses her policy instruments to maximize the sum of the expected welfare of consumer N and consumer D, subject to ensuring non-negative expected profit for the VIP. The regulator's policy instruments are the retail prices for electricity, the terms of the DG compensation the VIP must deliver to consumer D, and the VIP's generating capacity.

Consumer N's welfare  $(U^N(\cdot))$  is the difference between the value he derives from the electricity he consumes and the amount he pays for the electricity. Formally, consumer N's expected welfare is, for  $t' \neq t$   $(t, t' \in \{L, H\})$ :

$$E\left\{U^{N}(\cdot)\right\} = \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[V_{t}^{N}(X_{t}^{N}(r_{Nt}(\theta_{s}), \mathbf{r_{Nt}}'(\theta_{t'}), \theta_{s}), \theta_{s}) - r_{Nt}(\theta_{s})X_{t}^{N}(\cdot)\right] dF_{t}(\theta_{s}) - R_{N}.$$

$$(1)$$

Consumer D's welfare  $(U^D(\cdot))$  is the sum of the value he derives from the electricity he consumes and the compensation he receives for installing DG capacity and producing electricity, less the amount he pays for the electricity he purchases from the VIP and his costs of DG investment and production. Formally, consumer D's expected welfare is, for  $t' \neq t$   $(t, t' \in \{L, H\})$ :

$$E\left\{U^{D}(\cdot)\right\} = \sum_{t \in \{L,H\}} \left[\int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[V_{t}^{D}(X_{t}^{D}(r_{Dt}(\theta_{s}), \mathbf{r_{Dt'}}(\theta_{t'}), \theta_{s}), \theta_{s}) - r_{Dt}(\theta_{s}) X_{t}^{D}(\cdot)\right] dF_{t}(\theta_{s}) + \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[w_{it}(\theta_{s}) Q_{t}^{i}(K_{Di}, \theta_{s}) + w_{nt}(\theta_{s}) Q_{t}^{n}(K_{Dn}, \theta_{s}) - C_{t}^{D}(Q_{t}^{n}(\cdot, \theta_{s}), K_{Dn})\right] dF_{t}(\theta_{s})\right]$$

<sup>&</sup>lt;sup>22</sup>For expositional ease, the ensuing discussion focuses on settings where  $T(\cdot)$  is increasing in each of its arguments. Section 7 explicitly considers settings where DG capacity investments reduce the VIP's T&D costs.

<sup>&</sup>lt;sup>23</sup>These losses are relatively small in practice (Parsons and Brinckerhoff, 2012; U.S. Energy Information Administration (EIA), 2014b). Explicit accounting for these variable costs would not affect the key qualitative conclusions reported below.

+ 
$$k_{0i} + k_{0n} + k_i K_{Di} + k_n K_{Dn} - C_D^K(K_{Di}, K_{Dn}) - R_D$$
,  
where  $Q_t^n(\cdot, \theta_s) = \arg \max_Q \left\{ w_{nt}(\theta_s) Q - C_t^D(Q, K_{Dn}) \right\}$ . (2)

The VIP's profit  $(\pi)$  is the revenue it secures from selling electricity to consumers Dand N, less the sum of: (i) the DG compensation it pays to consumer D; (ii) the cost of its generating capacity; (iii) its cost of generating electricity; and (iv) its T&D costs. Formally, the VIP's expected profit is:

$$E\{\pi\} = \sum_{t \in \{L,H\}} \left[ \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ r_{Dt}(\theta_{s}) X_{t}^{D}(\cdot) + r_{Nt}(\theta_{s}) X_{t}^{N}(\cdot) - w_{it}(\theta_{s}) Q_{t}^{i}(K_{Di}, \theta_{s}) - w_{nt}(\theta_{s}) Q_{t}^{n}(K_{Dn}, \theta_{s}) - C_{t}^{G}(Q_{t}^{v}(\cdot), K_{G}) \right] dF_{t}(\theta_{s}) + R_{D} + R_{N} - k_{0i} - k_{0n} - k_{i} K_{Di} - k_{n} K_{Dn} - C^{K}(K_{G}) - T(K_{G}, K_{Di}, K_{Dn}) .$$
(3)

The timing in the model is as follows. The regulator first sets her policy instruments. Consumer D then chooses his DG capacity investments. Finally, the state is realized, DG production occurs, and the VIP supplies the realized demand for electricity and provides all required T&D of electricity in each demand period.<sup>24</sup>

# 3 State-Specific Pricing and DG Compensation

We first examine the optimal regulatory policy in a setting where smart meters are deployed ubiquitously and the regulator can set prices and DG payments that vary with the realization of the state variable in each period. The regulator's problem in this setting,  $[\text{RP}-\theta]$ , is:

$$\underset{r_{jt}(\theta_s), R_j, k_{0y}, k_{yt}, w_{yt}(\theta_s), K_G}{Maximize} \quad E\left\{U^N(\cdot)\right\} + E\left\{U^D(\cdot)\right\}$$
(4)

subject to: 
$$E\{\pi\} \ge 0$$
, (5)

where  $E \{ U^N(\cdot) \}$ ,  $E \{ U^D(\cdot) \}$ , and  $E \{ \pi \}$  are as defined in (1), (2), and (3), respectively.

Proposition 1 characterizes the key features of the optimal regulatory policy in this setting.

<sup>&</sup>lt;sup>24</sup>The sequencing of the two demand periods does not affect our findings.

**Proposition 1.** At a solution to  $[RP-\theta]$ : (i)  $\sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left| \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial K_G} \right| dF_t(\theta_s) = C^{K'}(K_G)$   $+ \frac{\partial T(\cdot)}{\partial K_G}$ ; (ii)  $r_{Nt}(\theta_s) = r_{Dt}(\theta_s) = w_{nt}(\theta_s) = w_{it}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)}$  for each  $t \in \{L,H\}$ and  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ ; and (iii)  $k_y = -\frac{\partial T(\cdot)}{\partial K_{Dy}}$  for  $y \in \{i, n\}$ .

Conclusion (i) in Proposition 1 indicates that the VIP's generating capacity is chosen to minimize the VIP's expected operating costs. In particular, this capacity is optimally increased to the point where the rate at which additional capacity reduces the VIP's expected generation costs is equal to the rate at which increased capacity increases the sum of the VIP's capacity costs and associated T&D costs.

Conclusion (ii) indicates that the unit retail price of electricity and the unit payment for DG production are both optimally set equal to the VIP's marginal cost of generating electricity in each state. These prices and payments ensure that consumers expand their consumption of electricity to the point where its marginal value is equal to its marginal cost of production in each state, regardless of the dispatachable (non-intermittent) source of production. Thus, the identified retail prices and DG payments ensure the efficient consumption and production of electricity, given installed generating capacities.<sup>25</sup>

Conclusion (iii) reports that the optimal unit DG capacity charge is the associated marginal increase in the VIP's T&D costs, which can vary by technology.<sup>26</sup> When he is compensated for the electricity he produces using the non-intermittent technology at a rate that reflects the VIP's marginal cost of generating electricity, consumer D will install the efficient level of  $K_{Dn}$  when he is required to bear the associated increment in the VIP's T&D costs.<sup>27</sup> In essence, this charge (and the associated payments for DG production) induce

<sup>&</sup>lt;sup>25</sup>The fixed retail charges  $(R_j)$  are set to ensure the VIP earns zero expected profit.

<sup>&</sup>lt;sup>26</sup>The increase in the VIP's T&D costs also can vary substantially across geographic regions. See Shlatz et al. (2013) and Cohen et al. (2015), for example.

<sup>&</sup>lt;sup>27</sup>The values of  $w_{it}(\theta_s)$  and  $k_i$  identified in Proposition 1 are not unique because the values of  $w_{it}(\theta_s)$  do not affect the amount of electricity produced by the installed intermittent capacity. If the  $w_{it}(\theta_s)$  payments are increased above  $\frac{\partial C_t^G}{\partial Q_t^w}$ , then  $k_i$  can be reduced accordingly to ensure consumer D installs the efficient level of intermittent DG capacity.

consumer D to account fully for the relevant expected social benefits and costs of investment in DG capacity.

The fact that the unit payment for DG production and the unit retail price of electricity are equated under the optimal policy identified in Proposition 1 has three important implications. First, net metering (as described in the Introduction) can be optimal for both DG technologies. Second, there is no strict gain from implementing DG compensation that varies by technology. Third, there is no strict gain from setting distinct retail prices for different consumers.<sup>28</sup> In this sense, when state-specific retail pricing and DG payments are feasible, relatively simple DG payment policies can secure the ideal outcome for consumers.

### 4 Externalities

The analysis to this point has abstracted from social losses due to externalities associated with the generation of electricity. To account for these losses, let  $S_t(Q_t^v, Q_t^i, Q_t^n)$  denote the social loss from environmental externalities in period t when the VIP produces output  $Q_t^v$ and when consumer D produces output  $Q_t^i$  with the intermittent technology and output  $Q_t^n$ with the non-intermittent technology.<sup>29</sup> Define  $S_{th}(\cdot) \equiv \frac{\partial S_t(\cdot)}{\partial Q_t^h} \geq 0$  for  $h \in \{v, i, n\}$ . The regulator's problem in this setting, [RP- $\theta e$ ], is to:

$$\underset{r_{jt}(\theta_s), R_j, k_{0y}, k_{yt}, w_{yt}(\theta_s), K_G}{Maximize} \quad E\left\{U^N(\cdot)\right\} + E\left\{U^D(\cdot)\right\} - E\left\{S_L(\cdot) + S_H(\cdot)\right\}$$
(6)

subject to: 
$$E\{\pi\} \ge 0$$
, (7)

where  $E \{ U^N(\cdot) \}$ ,  $E \{ U^D(\cdot) \}$ , and  $E \{ \pi \}$  are as defined in (1), (2), and (3), respectively.

**Proposition 2.** At a solution to  $[RP-\theta e]$ , conclusions (i) and (iii) in Proposition 1 hold. In addition,  $r_{Nt}(\theta_s) = r_{Dt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)} + S_{tv}(\cdot)$  and  $w_{yt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)} + S_{tv}(\cdot) - S_{ty}(\cdot)$  for all  $t \in \{L, H\}$ ,  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ , and  $y \in \{i, n\}$ .

<sup>&</sup>lt;sup>28</sup>Retail charges that differ across consumers can be optimal if the regulator values the welfare of consumer N differently from the welfare of consumer D.

<sup>&</sup>lt;sup>29</sup>For simplicity, we assume the social loss from externalities in one period does not vary with outputs produced in the other period. We also assume the total social loss from externalities is the sum of the corresponding losses in each period.

Proposition 2 reports that in order to induce efficient electricity consumption in the presence of externalities, it is optimal to raise the unit retail price of electricity above the VIP's marginal cost of production by the marginal social loss from externalities associated with the VIP's production of electricity  $(S_{tv}(\cdot))$ . Further, to induce consumer D to produce the efficient amount of electricity using the non-intermittent technology, it is optimal to increase the corresponding unit DG payment  $(w_{nt}(\theta_s))$  above the VIP's marginal cost of production by the difference between the marginal social loss from externalities due to production by the VIP and by consumer  $D(S_{tv}(\cdot) - S_{tn}(\cdot))$ . Thus, DG compensation is increased to the extent that the DG generation technology is "cleaner" than the VIP's generation technology, *ceteris paribus*.<sup>30</sup>

Proposition 2 implies that net metering generally is optimal only for electricity produced using a DG technology that generates no social losses from externalities. When retail prices are increased above the VIP's marginal cost by  $S_{tv}(\cdot)$  and unit DG payments are increased above the VIP's marginal cost by  $S_{tv}(\cdot) - S_{ty}(\cdot)$ , the retail prices and DG payments will coincide only if  $S_{ty}(\cdot) = 0$ . Thus, net metering may be optimal for solar DG production if it generates no social losses from externalities. In contrast, net metering generally is not optimal for DG production using CHP units, for example. Furthermore, even abstracting from differences in generating costs, optimal retail prices typically will vary with the VIP's generating technology. In addition, optimal DG payments typically will vary with the generating technologies employed by both the VIP and consumer D.

# 5 TOU Pricing and TOP DG Compensation

In practice, smart meters are not always deployed and employed ubiquitously. Consequently, state-specific retail prices and DG payments are not always feasible. In light of this fact, we now consider the setting where the regulator implements time of use (TOU) retail prices and time of production (TOP) DG payments (in addition to DG capacity charges).

<sup>&</sup>lt;sup>30</sup>Again, the values of  $w_{it}(\theta_s)$  and  $k_i$  identified in Proposition 2 are not unique because the  $w_{it}(\theta_s)$  payments do not affect consumer D's production of electricity using the intermittent DG technology, given installed capacity  $K_{Di}$ .

In this setting, the regulator can specify four distinct unit retail prices and four distinct DG payments. The retail prices can vary by customer and by demand period, but cannot vary within a period. The DG payments can vary by DG technology and by demand period, but cannot vary within a demand period. Formally, when only TOU retail prices and TOP DG payments are feasible, for  $y \in \{i, n\}$  and  $t \in \{L, H\}$ :

$$r_{jt}(\theta_s) = r_{jt}(\theta_{s'}) \equiv r_{jt} \text{ and } w_{yt}(\theta_s) = w_{yt}(\theta_{s'}) \equiv w_{yt} \text{ for all } \theta_s, \theta_{s'} \in [\underline{\theta}_t, \overline{\theta}_t].$$
 (8)

[RP-t] will denote the regulator's problem in this setting with TOU retail pricing and TOP DG payments. Formally, [RP-t] is problem [RP- $\theta$ ] with the additional restrictions specified in condition (8). Thus, for expositional clarity, we abstract from social losses from externalities.<sup>31</sup> Proposition 3 characterizes the key features of the optimal regulatory policy in this setting.

**Proposition 3.** At the solution to [RP-t]:

$$\begin{aligned} (i) \quad \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\underline{\theta}_{t}} \left| \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial K_{G}} \right| dF_{t}(\theta_{s}) &= C^{K\prime}(K_{G}) + \frac{\partial T(\cdot)}{\partial K_{G}}; \\ (ii) \quad \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ r_{jt} - \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} \right] \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}} dF_{t}(\theta_{s}) \\ &+ \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ r_{jt'} - \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s'}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s'})} \right] \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}} dF_{t'}(\theta_{s'}) = 0 \text{ for } t' \neq t, t, t' \in \{L,H\}, \\ (iii) \quad w_{nt} = \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} dF_{t}(\theta_{s}) \text{ for } t \in \{L,H\}; \text{ and} \\ (iv) \quad k_{y} = \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} - w_{yt} \right] \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} dF_{t}(\theta_{s}) - \frac{\partial T(\cdot)}{\partial K_{Dy}} \\ for \quad y \in \{i,n\}. \end{aligned}$$

Conclusion (i) in Proposition 3 again indicates that the VIP's generation capacity  $(K_G)$ 

<sup>&</sup>lt;sup>31</sup>Proposition 6 in the Appendix identifies how social losses from externalities affect the optimal regulatory policy in the present setting.

is chosen to minimize the VIP's expected operating costs, which include generation costs, capacity costs, and T&D costs.

Conclusion (ii) reports that the unit retail prices for electricity are set so that expected weighted deviations of prices from the incumbent's marginal cost of generating electricity are zero. Deviations of price from marginal cost are weighted more heavily when consumer demand is more sensitive to price, as under Ramsey pricing (Ramsey, 1927; Baumol and Bradford, 1970). Because price sensitivity can vary between consumers and across demand periods, the regulator typically is able to secure a strictly higher level of expected consumer welfare when she is able to set prices and DG payments that vary across consumers and across demand periods.

Conclusion (iii) implies that the unit compensation for non-intermittent DG production in period t is optimally set equal to the VIP's expected marginal cost of generating electricity in period t. By rewarding consumer D for increased DG production at a rate that reflects the associated expected reduction in the VIP's generation costs, the regulator encourages consumer D to expand his production of electricity from the non-intermittent DG technology to the level that minimizes expected industry generation costs, given installed generation capacities.

Conclusion (iv) indicates that the unit compensation  $(k_n)$  for non-intermittent DG investment  $(K_{Dn})$  is optimally set to induce consumer D to choose the level of  $K_{Dn}$  that minimizes expected industry costs, given  $K_G$  and  $K_{Di}$ . This is accomplished by setting  $k_n$  to equate consumer D's expected marginal benefit from increasing  $K_{Di}$  (i.e.,  $k_n + \sum_{t \in \{L,H\}} w_{nt} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \frac{\partial Q_t^n(\cdot,\theta_s)}{\partial K_{Dn}} dF_t(\theta_s)$ ) with the corresponding expected marginal net reduction in the VIP's costs, which is the difference between the expected marginal reduction in the VIP's generating costs  $(\sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} | \frac{\partial Q_t^v}{\partial K_{Dn}} | dF_t(\theta_s))$  and the marginal increase in the VIP's T&D costs  $(\frac{\partial T(\cdot)}{\partial K_{Dn}})$ .

Conclusions (iii) and (iv) together imply that the unit capacity charge  $(-k_n)$  is set to induce the desired capacity investment in the non-intermittent DG technology  $(K_{Dn})$ , given DG payments ( $w_{nL}$  and  $w_{nH}$ ), whereas these payments are set to induce the desired level of electricity production, given the installed capacity investment ( $K_{Dn}$ ).

Conclusion (iv) also reports that the compensation consumer D anticipates for installing intermittent DG capacity  $(K_{Di})$  and producing electricity using this capacity is set to ensure the consumer chooses the level of  $K_{Di}$  that minimizes expected industry production costs, given  $K_G$  and  $K_{Dn}$ .<sup>32</sup> Because consumer D cannot control the amount of electricity produced by each unit of capacity of the intermittent technology, the corresponding unit DG compensation levels ( $w_{iL}$  and  $w_{iH}$ ) do not affect electricity production, given installed capacity  $K_{Di}$ . Consequently, the regulator has considerable leeway in setting  $w_{iL}$ ,  $w_{iH}$ , and  $k_i$  to jointly induce the desired  $K_{Di}$  capacity investment. In particular, the regulator can set  $w_{iL} = w_{nL}$  and  $w_{iH} = w_{nH}$  and adjust  $k_i$  to induce the desired level of  $K_{Di}$ . Therefore, the regulator derives no strict gain from an ability to set technology-specific payments for DG production of electricity in this setting with TOU retail prices and TOP DG compensation.<sup>33</sup>

#### 6 Settings with More Restricted Policy Instruments

The preceding analysis assumes the regulator can set retail prices that differ across consumers and across demand periods. It also assumes the regulator can both set payments for DG production of electricity and specify compensation (or charges) for DG capacity investment. We now briefly illustrate how the optimal policy changes when the regulator's policy instruments are more limited.

### 6.1 No Customer-Specific Pricing

First consider the optimal regulatory policy when the regulator can set: (i) a single fixed retail charge for electricity (R) for both consumers; (ii) a single unit retail price (r)of electricity that the two consumers face throughout both demand periods; (iii) a single technology-specific unit payment for DG production  $(w_y \text{ for } y \in \{i, n\})$  that prevails in both

<sup>&</sup>lt;sup>32</sup>Because the VIP generates the difference between the quantity of electricity demanded and the quantity supplied by consumer D,  $\frac{\partial Q_t^v}{\partial Q_t^i} = -1$  and so  $\frac{\partial Q_t^v}{\partial Q_t^i} \frac{\partial Q_t^i(\cdot, \theta_s)}{\partial K_{Di}} = -\mu_t \theta_s$ .

<sup>&</sup>lt;sup>33</sup>This conclusion does not hold if there is more than one DG technology for which consumer D can exercise non-trivial control over the amount of DG production, given the prevailing capacity investments.

demand periods; and (iv) technology-specific unit compensation (or charges) for DG capacity  $(k_i \text{ and } k_n)$ . Let [RP] denote the regulator's problem in this setting.

**Proposition 4.** At a solution to 
$$[RP]$$
: (i)  $\sum_{j \in \{D,N\}} \sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left[ r - \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)} \right]$   
$$\frac{\partial X_t^j(\cdot,\theta_s)}{\partial r} dF_t(\theta_s) = 0; \quad (ii) \ w_n = \sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)} dF_t(\theta_s); \text{ and } (iii) \ k_y =$$
$$\sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left[ \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s),K_G)}{\partial Q_t^v(\cdot,\theta_s)} - w_y \right] \frac{\partial Q_t^y(K_{Dy},\theta_s)}{\partial K_{Dy}} dF_t(\theta_s) - \frac{\partial T(\cdot)}{\partial K_{Dy}} \text{ for } y \in \{i,n\}.^{34}$$

A comparison of Propositions 3 and 4 reveals that similar considerations underlie the optimal regulatory policy regardless of the regulator's ability to set different prices for different customers or in different demand periods. Conclusion (i) in Proposition 4 indicates that the regulator optimally sets the constant unit retail price of electricity (r) so that the expected weighted deviations between r and the VIP's marginal cost of generating electricity are zero. The weights again reflect the sensitivity of consumer demand to price (for both consumers and in both demand periods).

Conclusion (ii) reports that the optimal unit payment for electricity produced with the non-intermittent technology is equal to the VIP's expected marginal cost of generating electricity (in both demand periods). Conclusion (iii) states that the optimal compensation for DG capacity investment ensures that consumer D's marginal expected return to increasing the capacity of each technology is equal to the corresponding marginal reduction in the VIP's costs (which include both generation costs and T&D costs).

Because the established value of  $w_i$  does not affect the amount of electricity produced from a given investment in the capacity of the intermittent technology  $(K_{Di})$ , the regulator can set  $w_i = r$  and adjust  $k_i$  accordingly to induce the optimal  $K_{Di}$  investment. Consequently, net metering for electricity produced with the intermittent DG technology can be optimal. In contrast, net metering for electricity produced with the non-intermittent DG technology generally is not optimal, even in the absence of social losses from externalities.

<sup>&</sup>lt;sup>34</sup>Conclusion (i) in Proposition 3 also holds and the VIP earns zero expected profit at the solution to [RP].

# 6.2 No DG Capacity Charges

Although DG capacity charges have been proposed in several jurisdictions, they are not presently deployed ubiquitously.<sup>35</sup> When a regulator is unable to implement capacity charges, she must employ payments for DG production to induce desired investment in DG capacity. Consequently, it typically is no longer optimal to set these payments to induce efficient levels of DG electricity production, given installed generating capacities. To illustrate this more general conclusion, consider the setting with state-specific retail prices and DG payments analyzed in section 3, with the exception that the regulator cannot impose any capacity charges (so  $k_{0y} = k_{yt}(\theta_s) = 0$  for all  $y \in \{i, n\}, t \in \{L, H\}$ , and  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ ). Let [RP- $\theta k$ ] denote the regulator's problem in this setting.

**Proposition 5.** At a solution to  $[RP-\theta k]$ , for all  $t \in \{L, H\}$  and  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ : (i)  $r_{Dt}(\theta_s) = r_{Nt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v(\cdot,\theta_s)}$ ; and (ii)  $w_{yt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v(\cdot,\theta_s)} - \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_s)}$  $\left[ \left( \frac{\partial Q_t^y(\cdot,\theta_s)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_s)} + \frac{\partial Q_t^y(\cdot,\theta_s)}{\partial w_{yt}(\theta_s)} \right) f_t(\theta_s) \right]^{-1}$  for  $y \in \{i, n\}$ .<sup>36</sup>

Conclusion (i) in Proposition 5 indicates that in order to induce the efficient consumption of electricity, the regulator sets the retail price of electricity equal to the VIP's marginal cost of generating electricity in each state in each demand period. To induce consumer D to produce the efficient level of electricity using the non-intermittent technology, the regulator would like to set each  $w_{nt}(\theta_s)$  payment at this same level. However, these payments would induce consumer D to install more than the efficient level of capacity of the non-intermittent technology in the absence of a DG capacity charge that reflects the increase in T&D costs the VIP experiences as  $K_{Dn}$  increases. Consequently, as conclusion (ii) in Proposition 5 indicates, the regulator optimally reduces the  $w_{nt}(\theta_s)$  payments below  $\frac{\partial C_t^{CG}(\cdot)}{\partial Q_t^n}$ . The amount of the reduction is proportional to the rate at which an increase in the DG payment increases

<sup>&</sup>lt;sup>35</sup>DG capacity charges have been proposed in Arizona, Hawaii, Kansas, Maine, New Mexico, Oklahoma, Utah, and Wisconsin (NCCETC, 2015d).

<sup>&</sup>lt;sup>36</sup>Conclusion (i) in Proposition 1 also holds and the VIP earns zero expected profit at the solution to [RP- $\theta k$ ].

the VIP's T&D costs per corresponding unit of increased DG output.<sup>37,38</sup>

Proposition 5 indicates that, in contrast to the setting where the regulator can set DG capacity charges, net metering  $(w_{yt}(\theta_t) = r_{yt}(\theta_t))$  is not optimal for either technology, even in the absence of social losses from externalities. Thus, although such relatively simple DG payments can be optimal when the regulator has access to a full array of policy instruments, these payments typically are not optimal when the regulator's instruments are more limited, and so she must employ DG payments both to induce desired DG capacity investment and to affect DG electricity production given the installed capacities.

# 7 Illustrating the Impact of Capacity Charges

We now present numerical solutions to illustrate how the ability to set capacity charges affects the optimal regulatory policy and the financial returns from undertaking DG. For simplicity, we assume there is a single demand period each day and consumer demand for electricity is iso-elastic,<sup>39</sup> so consumer j's demand for electricity in state  $\theta$  with retail price r is  $X^{j}(r, \theta) = m_{j} [1 + \theta] r^{\alpha_{j}}$ , where  $m_{j} > 0$  and  $\alpha_{j} < 0$  for  $j \in \{D, N\}$ .<sup>40</sup> We assume the price

<sup>&</sup>lt;sup>37</sup>The numerator of the last term in conclusion (ii) in Proposition 5 for y = n is the rate at which the VIP's T&D costs increases as  $w_{nt}(\theta_s)$  increases due to the induced increase in  $K_{Di}$ . The denominator of this term reflects the rate at which output produced using the non-intermittent DG technology  $(Q_t^n(\cdot))$  increases as  $w_{nt}(\theta_s)$  increases. The increase in  $Q_t^n(\cdot)$  arises from two sources. First, the increase in  $w_{nt}(\theta_s)$  induces consumer D to increase  $K_{Dn}$ , which reduces the consumer's marginal cost of producing electricity, which in turn induces increased production. Second, the increase in  $w_{nt}(\theta_s)$  increases the prevailing rate of compensation for electricity production, which induces increased production. The corresponding interpretation when y = i is analogous except that an increase in  $w_{it}(\theta_s)$  only affects the amount of electricity produced using the intermittent DG technology by inducing consumer D to increase  $K_{Di}$  (i.e.,  $\frac{\partial Q_t^i(\cdot, \theta_s)}{\partial w_{it}(\theta_s)} = 0$ ).

<sup>&</sup>lt;sup>38</sup>The central considerations that underlie the findings in Proposition 5 persist in other settings where the regulator's instruments are more limited (e.g., where she can only set TOU retail prices and TOP DG payments).

<sup>&</sup>lt;sup>39</sup>We also assume the regulator cannot impose fixed, state-specific, or customer-specific retail charges. She also cannot implement state-specific DG capacity charges or fixed DG charges that do not vary with the level of installed capacity. The regulator seeks to maximize expected consumer welfare while ensuring nonnegative expected profit for the VIP.

<sup>&</sup>lt;sup>40</sup>This relationship is assumed to hold for  $r \leq r^m$ . We assume  $X^j(\cdot) = 0$  for  $r > r^m$  to ensure finite values for  $E\{U^j\}, j \in \{D, N\}$ . We set  $r^m = 800$ , reflecting particularly high estimates of customer valuations of lost load (London Economics International, 2013). Results similar to those reported below hold if the more general iso-elastic demand function  $X^j(r, \theta) = m_j [1 + \theta^s] r^{\alpha_j}$  is employed, where s differs from 1.

elasticity of demand for electricity is -0.25 for both consumers, so  $\alpha_D = \alpha_N = -0.25$ .<sup>41</sup>

We begin by considering a "baseline" setting in which the VIP primarily employs noncoal resources to serve a relatively large number of customers, as in California, for example.<sup>42</sup> In this setting, the demand parameters  $m_D$  and  $m_N$  are chosen to equate the equilibrium expected demand in the model with  $\overline{X} = 25,391$  (MWh), the average hourly consumption of electricity in California in 2014.<sup>43</sup> Formally,  $m_j$  is chosen to ensure  $E\{m_j [1+\theta] \tilde{r}^{\alpha_j}\} =$  $\eta_j \overline{X}$  for  $j \in \{D, N\}$ , where  $\eta_D$  denotes the fraction of demand accounted for by customers who undertake some distributed generation of electricity,  $\eta_N = 1 - \eta_D$ , and  $\tilde{r} = 143.8165$ reflects the average unit retail price of electricity (\$/MWh) in California in 2014 (California Public Utilities Commission, 2015). We initially assume  $\eta_D = 0.1$  to reflect the potential deployment of PV panels in the U.S. in the near future.<sup>44</sup>

Recall that the distribution of the state variable ( $\theta$ ) reflects variation in the production of electricity from installed DG capacity. To specify this distribution, we first plot the ratio of the MW's of electricity produced by photo-voltaic (PV) panels to the year-end installed generating capacity ( $\overline{K}_D = 3,254$  MW) of PV panels in California for each of the 8,760 hours in 2014.<sup>45</sup> We then employ maximum likelihood estimation to fit a density function to

<sup>&</sup>lt;sup>41</sup>Estimates of the short-run price elasticity of demand for electricity for residential consumers range from -0.13 (Paul et al., 2009) to -0.35 (Espey and Espey, 2004). Corresponding long-run estimates reflect more elastic demand (e.g., between -0.40 (Paul, 2009) and -0.85 (Espey and Espey, 2004)). Commercial and industrial customers typically exhibit less elastic demands for electricity (e.g., Wade, 2003; Taylor et al., 2005; Paul et al., 2009). Brown and Sappington (2016) demonstrate how the optimal regulatory policy changes as this parameter and other parameters change. The qualitative results presented in this section generally are robust to considerable variation in parameter values.

<sup>&</sup>lt;sup>42</sup>The ensuing calculations are not intended to characterize actual or likely outcomes in California or any other specific jurisdiction However, the California data is useful in modeling a setting where a VIP generates a relatively large amount of electricity employing primarily non-coal resources.

 $<sup>{}^{43}\</sup>overline{X}$  is the sum of: (i)  $\overline{Q}_v = 24,577$  MWh, the average amount of electricity sold hourly by California utilities in 2014 (California Independent System Operator (CAISO), 2015a); and (ii) the estimated average hourly electricity generated from solar DG in California in 2014. This latter estimate is 25.0% of  $\overline{K}_D$ , the 3,254 MW of PV capacity installed in California at year end 2014 (California Solar Statistics, 2015). The 25.0% reflects the utilization rate of solar DG capacity in California (CAISO, 2015b).

<sup>&</sup>lt;sup>44</sup>10.6% of consumers undertook some DG of electricity in Hawaii in 2014. The corresponding percentages are 2% in California and 1.6% in Arizona (EIA, 2015b). Schneider and Sargent (2014) report rapid growth in the installation of solar panels in recent years.

 $<sup>^{45}</sup>$ The data on PV output are derived from CAISO (2015b).

the 4,443 (49.4%) of the observations that are strictly positive. Standard tests reveal that the beta distribution with parameters (1.165, 1.204855) fits the data well, and so is used as the density function for  $\theta$  in the ensuing analysis.<sup>46</sup>

The VIP's capacity costs are taken to be quadratic, i.e.,  $C^K(K_G) = a_K K_G + b_K (K_G)^2$ . Estimates of the cost of the generation capacity required to produce a MWh of electricity range from \$16.1/MWh for a conventional combined cycle natural gas unit to \$81.9/MWh for a nuclear facility (EIA, 2015a). We set  $a_K$  at the lower bound of this range (16.1). We also set  $b_K = 0.00045$  to ensure that the marginal cost of capacity required to generate a MWh of electricity is \$81.9 at the observed level of centralized non-renewable generation capacity in California in 2014 ( $\overline{K}_G = 72,926$  MW) (California Energy Commission, 2015).

For simplicity, the VIP's T&D costs are assumed to be linear, i.e.,  $T(K_G, K_{Di}, K_{Dn}) = a_T^G K_G + a_T^{Di} K_{Di} + a_T^{Dn} K_{Dn}$ . Estimates of these costs vary widely. EIA (2015a) estimates utility transmission capacity costs associated with generating a MWh of electricity are between \$1.2 and \$3.5 for centralized, non-renewable generation, and between \$4.1 and \$6.0 for PV generation. Reflecting the midpoints of these cost estimates, we initially assume  $a_T^G = 2.35$  and  $a_T^{Di} = a_T^{Dn} = 5.05$ .<sup>47</sup>

We assume the VIP's cost of generating  $Q^v$  units of electricity when it has  $K_G$  units of capacity to be  $C^G(Q^v, K_G) = \left[a_v + \frac{c_v}{K_G}\right]Q^v + b_v (Q^v)^2$ , where  $a_v$ ,  $b_v$ , and  $c_v$  are positive constants. Thus, increased capacity reduces the VIP's cost of generating electricity at a diminishing rate.<sup>48</sup> We set  $b_v = 0.003$  and  $a_v + \frac{c_v}{K_G} = 28.53$ , reflecting Bushnell (2007)'s estimates.<sup>49</sup>  $c_v$  is chosen to equate the observed marginal benefit  $\left(\frac{c_v \overline{Q}^v}{(\overline{K_G})^2}\right)$  and marginal cost

<sup>48</sup>Formally,  $\frac{\partial C^G(\cdot)}{\partial K_G} = -\frac{c_v Q^v}{(K_G)^2} < 0$  and  $\frac{\partial^2 C^G(\cdot)}{\partial (K_G)^2} = \frac{2 c_v Q^v}{(K_G)^3} > 0$ , which implies  $\frac{\partial}{\partial K_G} \left| \frac{\partial C^G(\cdot)}{\partial K_G} \right| < 0$ .

 $<sup>^{46}</sup>$ The chi-squared, Kolmogorov-Smirnov, and Anderson-Darling tests also reveal that the generalized extreme value (GEV) distribution with parameter values (0.4827, 0.3088, -0.7135) fits the data reasonably well. Findings very similar to those reported below arise when this GEV distribution replaces the identified beta distribution.

<sup>&</sup>lt;sup>47</sup>Alternative estimates for these parameters are considered below.

<sup>&</sup>lt;sup>49</sup>Employing a cost function of the form  $C(Q^v) = a Q^v + b(Q^v)^2$ , Bushnell (2007) estimates a = 28.53 and b = 0.003.

 $(a_K + 2 b_K \overline{K}_G + a_T^G)$  of VIP capacity.<sup>50</sup>

The cost of installing  $K_{Di}$  units of intermittent DG capacity and  $K_{Dn}$  units of nonintermittent DG capacity is assumed to be  $C_D^K(K_{Di}, K_{Dn}) = a_{Di} K_{Di} + b_{Di} (K_{Di})^2 + a_{Dn} K_{Dn} + b_{Dn} (K_{Dn})^2$ . Estimates of the unsubsidized cost of residential PV capacity vary between \$100 and \$400/MWh (Branker et al., 2011; EIA, 2015a). When the 30 percent federal income tax credit (ITC) is applied, these estimates decline to between \$70 and \$280/MWh. State subsidies further reduce these estimates to between \$45 and \$255/MWh (NCCETC, 2015a,b,c). We initially set  $a_{Di} = 150$ , the midpoint of this lattermost range. We also set  $b_D = 0.0038$ to ensure ensure that the marginal cost of DG capacity when  $K_{Di} = \overline{K}_D = 3,254$  (i.e.,  $a_D + 2 b_D \overline{K}_D$ ) is 175, the midpoint of the range of estimated costs after applying the ITC.

Estimates of the cost of DG capacity using natural gas reciprocating engine and gas turbines range from \$34.09 to \$52.81/MWh (PacifiCorp, 2013). We set  $a_{Dn}$  equal to the lower bound of this range (34.09) and select  $b_{Dn}$  to ensure the marginal cost of capacity at the level of combined heat and power (CHP) DG in California,  $\overline{K}_{Dn} = 8,518$  MW (ICF, 2012), reflects the upper bound of this range, so  $a_{Dn} + 2b_{Dn}\overline{K}_{Dn} \approx 52.81$ .

Finally, we assume there are no social losses from externalities due to (solar) distributed generation of electricity. The corresponding losses are assumed to increase linearly with electricity produced by the VIP.<sup>51</sup> Specifically, we assume  $L(Q^v, Q^D) = e_v Q^v$ , where  $e_v = \phi_c e_c + \phi_g e_g + \phi_o e_o$ .  $\phi_c$ ,  $\phi_g$ , and  $\phi_o$  denote the fraction of the VIP's electricity production that is generated by coal, natural gas, and other units, respectively.<sup>52</sup> We initially set  $e_c = 37.231$ ,  $e_g = 21.029$ , and  $e_o = 0$ .  $e_j$  is the estimated unit loss from environmental externalities for technology  $j \in \{c, g, o\}$ . This estimate is the product of \$38, the estimated social cost of a metric ton of CO<sub>2</sub> emissions (EPA, 2013), and the metric tons of CO<sub>2</sub>

<sup>&</sup>lt;sup>50</sup>Recall that  $\overline{Q}^v = 24,577$  is the average MWh's of electricity sold daily by California utilities in 2014 and  $\overline{K}_G = 72,926$  is the MW of centralized non-renewable generation capacity in California at year-end 2014.

Thus, the initial value for  $c_v$  (and hence  $a_v$ ) reflects the assumption that the welfare-maximizing level of capacity in the model is  $\overline{K}_G$ .

<sup>&</sup>lt;sup>51</sup>For simplicity, we abstract from the non-linearities that arise in practice as different technologies are employed to meet baseload, systematic non-baseload, and peak-load demand for electricity.

<sup>&</sup>lt;sup>52</sup>In practice, other units primarily reflect hydro and nuclear production.  $\phi_o = 1 - \phi_c - \phi_g$ .

emissions that arise when technology j is employed to produce a MWh of electricity.<sup>53</sup> In the current baseline setting, we assume  $\phi_c = 0.064$  and  $\phi_g = 0.445$ , reflecting the fraction of electricity generated by California utilities in 2014 using coal and natural gas generating units, respectively (California Energy Commission, 2015). Consequently,  $e_v = 11.746$ .

Table 1 characterizes the optimal regulatory policy and industry outcomes in the baseline setting both when the regulator can set DG capacity charges (column 3) and when she cannot do so (column 2).<sup>54</sup>  $\pi_{Dy}$  in Table 1 denotes the expected profit from DG employing technology  $y \in \{i, n\}$ , where *i* denotes "intermittent" and *n* denotes "non-intermittent."<sup>55</sup> E {S} denotes the expected social losses due to environmental externalities. Because consumer *D* cannot control the level of DG output from the intermittent technology other than through his choice of capacity  $(K_{Di})$ ,  $w_i$  and  $k_i$  are redundant instruments for the regulator when both are available. In this event, we assume the regulator implements net metering  $(w_i = r)$  for the intermittent DG ("DG-*i*") technology.

Four elements of Table 1 warrant emphasis. First, when DG capacity charges are feasible, the regulator optimally imposes a capacity tax ( $k_i < 0$ ) on the DG-*i* technology. The tax reduces investment in the technology below the (socially excessive) level that net metering would induce in the absence of a tax. When DG capacity charges are not feasible, the unit payment for DG-*i* output ( $w_i$ ) is reduced substantially below the unit retail price of electricity (r) in order to limit excessive investment in DG-*i* capacity.

Second, the regulator also imposes a capacity tax on the non-intermittent DG ("DG-n") technology when capacity charges are feasible. Doing so limits excessive investment in DG-n capacity that otherwise arises when  $w_n$  is set both to induce an efficient level of

<sup>&</sup>lt;sup>53</sup>EIA (2014a) estimates that 2.16 (1.22) pounds of carbon dioxide are emitted when a KWh of electricity is produced using a coal (natural gas) generating unit. These estimates are multiplied by 1,000 to convert KWhs to MWhs, and divided by 2,204.62 to convert pounds to metric tons. Thus,  $e_c = 38 [2.16] \frac{1,000}{2,204.62} = 37.231$  and  $e_g = 38 [1.22] \frac{1,000}{2,204.62} = 21.029$ .

<sup>&</sup>lt;sup>54</sup>Brown and Sappington (2016) review the methodology employed to derive the findings in Table 1.

<sup>&</sup>lt;sup>55</sup>Profit is the difference between: (i) the sum of payments for DG output and any relevant payments for installing DG capacity; and (ii) the sum of DG capacity costs and any relevant variable costs of DG production.

DG-*n* output and motivate investment in DG-*n* capacity. The reduction in DG-*n* capacity investment that is optimally induced when DG capacity charges are feasible is offset in part by increased centralized generating capacity  $(K_G)$ .

Third, the significant (12.73%) decline in DG-*n* capacity when capacity charges are feasible results in a substantial (23.83%) decline in profit from DG-*n* operations, which reduces the expected utility of consumer D.<sup>56</sup> Fourth, because electricity production is shifted from the relatively "dirty" DG-*n* technology to the relatively "clean" centralized production technology, the expected social losses from environmental externalities decline when capacity charges are feasible.

Outcomes in the baseline setting are sensitive to the presumed impact of DG on the utility's T&D costs. Distinct outcomes can arise if DG reduces, rather than increases, these costs. To illustrate this conclusion, suppose all parameters are as specified in the baseline setting except  $\alpha_T^{Dn}$  is reduced to -11.0, so investment in the DG-*n* technology reduces the utility's T&D costs substantially.<sup>57</sup> As Table 2 reports, the regulator now implements a capacity payment ( $k_n > 0$ ) rather than a capacity tax for the DG-*n* technology when capacity charges are feasible. Perhaps more surprisingly, the regulator also increases the unit payment for DG output ( $w_n$ ). The capacity payment induces a substantial increase in DG-*n* capacity, and a corresponding reduction in centralized capacity ( $K_G$ ) is implemented. The reduction in  $K_G$  increases the VIP's marginal cost of producing electricity. In light of this higher marginal cost,  $w_n$  is increased to induce more output from the DG-*n* technology, which admits a reduction in (now relatively costly) output by the VIP.<sup>58</sup>

 $<sup>^{56}</sup>$ Consumer N's expected utility increases because the unit retail price of electricity declines when the regulator can employ capacity charges to reduce industry production costs. Aggregate expected wefare is little changed, in part because consumer surplus substantially exceeds industry costs under the presumed demand and cost structures.

<sup>&</sup>lt;sup>57</sup>Cohen et al. (2015) report widely varying estimates of the impact of DG on utility T&D costs. Clean Power Research (2014) estimates that these costs decline by \$11.00 for each MWh of PV capacity investment.

<sup>&</sup>lt;sup>58</sup>If  $\alpha_T^{Di} = \alpha_T^{Dn} = -11.0$ , then when capacity charges become feasible, the regulator reduces  $w_n$  as she implements a capacity payment,  $k_n > 0$ . This is the case because the reduction in the VIP's capacity is less pronounced in this setting than in the setting of Table 2, so the corresponding increase in the VIP's marginal cost of producing electricity is less pronounced.

The increased investment in DG-n capacity in this setting when capacity charges are feasible leads to a pronounced (106.73%) increase in the profit from DG-n operations. The expected social losses from externalities increase as output is shifted from the VIP's relatively clean technology to the DG-n technology.

To illustrate the different qualitative conclusions that can arise under other circumstances, now consider a "coal-intensive" setting where the VIP primarily employs coal-powered generating units to serve a relatively small market. Specifically, suppose  $\phi_c = 0.9$ ,  $\phi_g = 0.1$ , and the demand parameters  $(m_j)$  are chosen to ensure  $E\{m_j [1 + \theta^{\beta_j}] \tilde{r}^{\alpha_j}\} = \eta_j \hat{X}$ , where  $\hat{X} = 21,404$ .<sup>59</sup> Further suppose the VIP's capacity parameters  $(a_K = 16.1, b_K = 0.000674)$ ensure that the marginal cost of capacity required to generate a MWh of electricity is approximately 60.4 when  $K_G = \hat{K}_G = 32,854$ .<sup>60,61</sup>

Table 3 reports that when capacity charges become feasible in the coal intensive setting, the regulator optimally implements a capacity payment  $(k_i > 0)$  for the clean DG-*i* technology and a capacity tax  $(k_n < 0)$  for the less clean DG-*n* technology. This payment and tax induce increased investment in DG-*i* capacity and reduced investment in DG-*n* capacity. On balance, profit from DG operations declines and social losses from externalities increase slightly.

Tables 1 - 3 illustrate that the ability to set capacity charges can induce significant changes in industry outcomes and that the nature and extent of the changes can vary substantially across settings. Brown and Sappington (2016) show that significant and varied changes persist in other plausible settings of interest.

<sup>&</sup>lt;sup>59</sup>The smaller level of expected demand  $(\widehat{X} < \overline{X})$  reflects the average daily consumption of electricity in Ohio in 2014 (EIA, 2015c).  $e_v = 35.692$  when  $\phi_c = 0.9$  and  $\phi_g = 0.1$ .

 $<sup>{}^{60}\</sup>hat{K}_G = 32,854$  MW reflects the level of centralized non-renewable generation capacity in Ohio in 2013 (EIA, 2015c). The estimated cost of capacity required to produce a MWh of electricity using a coal generating unit is \$60.4 (EIA, 2015a).

<sup>&</sup>lt;sup>61</sup>Much as in the baseline setting, we choose  $a_v$ ,  $b_v$ , and  $c_v$  to reflect Bushnell (2007)'s estimates for the serving region of the PJM regional transmission organization (i.e., a = 0 and b = 0.0009), assuming that the welfare-maximizing level of VIP capacity in the model is  $\hat{K}_G$ . The values of  $a_n$  and  $b_n$  are set as in the baseline setting, except that the benchmark level of DG-n capacity is reduced to  $\hat{K}_{Dn} = 3,837.35 = 0.1168 \cdot \hat{K}_G$ , reflecting the same ratio of DG-n to VIP capacity ( $\overline{K}_{Dn}/\overline{K}_G = 0.1168$ ) as in the baseline setting.

### 8 Conclusions

We have examined the optimal design of policies to promote efficient distributed generation (DG) of electricity. Our analysis produced the following seven conclusions that may help to inform the current debate about desirable DG policies.

First, there is no single policy that best promotes efficient DG in all settings. The optimal policy varies with the characteristics of the prevailing environment, including the set of policy instruments available to the regulator and the relevant DG production technologies.

Second, DG capacity charges enhance consumer welfare in the presence of non-intermittent DG production technologies. The capacity charges can be employed to induce efficient investment in DG capacity while payments for electricity produced with the non-intermittent DG technology can be employed to induce efficient production levels.

Third, in the absence of social losses from externalities, retail prices and DG payments for electricity that reflect the VIP's marginal cost of generating electricity induce both the efficient consumption of electricity and the efficient allocation of electricity production among industry suppliers. Therefore, common net metering policies (whereby the unit payment for distributed production of electricity is set equal to the prevailing retail price of electricity) can be part of an optimal policy. This is the case as long as DG capacity charges are set to reflect the impact of increased DG capacity on the VIP's transmission and distribution costs.

Fourth, when state-specific (e.g., real time) prices and DG payments are not feasible but DG capacity charges are feasible, the optimal unit DG payment for electricity reflects the VIP's expected marginal cost of generating electricity. In contrast, the optimal unit retail price of electricity is set so that a weighted average of deviations between price and the VIP's marginal cost of generating electricity is zero. Consequently, net metering typically is not optimal for non-intermittent DG technologies. Net metering can be optimal for intermittent DG technologies because DG payments do not affect the amount of electricity produced with the installed capacity of such technologies. Therefore, DG capacity charges can be set to induce efficient intermittent capacity investment, given unit DG payments that reflect prevailing retail prices of electricity.

Fifth, net metering typically is not optimal even for intermittent DG technologies when the regulator is unable to set DG capacity charges. In this case, unit DG payments for electricity must be designed to induce efficient DG capacity investment whereas retail prices optimally are designed to induce efficient electricity consumption.

Sixth, fixed retail charges for electricity can play an important role in inducing efficient electricity consumption, particularly when they do not affect the demand for electricity. Such fixed charges enable the regulator to set unit retail electricity prices to reflect the VIP's marginal cost of production while ensuring a normal expected profit for the VIP.<sup>62</sup>

Seventh, the presence of social losses from externalities affects optimal retail prices and DG payments in intuitive ways. In particular, retail prices of electricity are increased above the VIP's marginal cost of generating electricity to reflect the marginal social loss from externalities associated with the VIP's electricity generation. In addition, the unit DG payment for electricity is increased above the VIP's marginal cost of generating electricity to the extent that losses from externalities decline as production is shifted from the VIP to distributed generators. These changes imply that common net metering policies typically are not optimal unless distributed generation of electricity entails no social losses from externalities.

In closing, we note that although we have considered a fairly broad set of policy instruments for the regulator, an even more comprehensive set of instruments warrants formal investigation. Additional instruments of potential interest include demand charges (e.g., Hledik, 2014; Faruqui and Hledik, 2015) and policies that promote demand response (e.g., Chao, 2011; Brown and Sappington, 2015b) and energy conservation (e.g., Brennan, 2010b; Chu and Sappington, 2013). Future research also might consider the changes that arise in the presence of limited consumer information and differential concerns about the welfare of different consumers (e.g., residential vs. industrial consumers or residential consumers who

<sup>&</sup>lt;sup>62</sup>Fixed DG capacity charges  $(k_{0i} \text{ and } k_{0n})$  also can serve this function if they do not induce consumers to forego efficient DG capacity investment.

	can	invest	$\mathrm{in}$	$\mathbf{DG}$	technologies	vs.	those	who	cannot	$).^{63}$
--	-----	--------	---------------	---------------	--------------	-----	-------	-----	--------	-----------

Variable	No DG Charges	DG Charges	Change	% Change
r	271.23	271.03	-0.20	-0.07
$w_i$	186.04	271.03	84.99	45.68
$w_n$	155.36	157.48	2.12	1.36
$k_i$	0	-20.64	N/A	N/A
$k_n$	0	- 3.19	N/A	N/A
K <sub>Di</sub>	4,742.54	4,744.53	1.99	0.04
$K_{Dn}$	4,777.62	4,169.66	-607.96	-12.73
$K_G$	63,838.65	64,378.74	540.09	0.85
$\pi_{Di}$	20,755.65	20,773.10	17.45	0.08
$\pi_{Dn}$	30,441.31	23, 186.81	-7,254.50	-23.83
$E\left\{U^{D}\right\}$	507, 287.91	500,475.09	-6,812.81	-1.34
$E\left\{U^{N}\right\}$	4, 104, 818.49	4,108,636.58	3,818.09	0.09
$E\left\{S\right\}$	273, 488.68	269,849.18	-3,639.50	- 1.33

 Table 1. Outcomes in the Baseline Setting.

<sup>&</sup>lt;sup>63</sup>It can be shown that the key qualitative conclusions drawn above continue to hold when the regulator seeks to maximize a weighted average of the welfare of consumer D and the welfare of consumer N, while ensuring a reservation level of expected welfare for both cosumers. In this setting, fixed retail charges are set to deliver a higher level of welfare to the favored consumer while ensuring the less-favored consumer secures his reservation welfare level. Binding limits on feasible fixed retail charges would introduce additional changes to the optimal unit retail prices of electricity.

Variable	No DG Charges	DG Charges	Change	% Change
r	268.07	270.69	2.62	0.98
$w_i$	187.51	270.69	83.17	44.36
$w_n$	164.72	166.41	1.69	1.02
$k_i$	0	-19.23	N/A	N/A
k <sub>n</sub>	0	7.42	N/A	N/A
K <sub>Di</sub>	4,936.16	5,179.84	243.68	4.94
K <sub>Dn</sub>	7,453.82	10,071.28	2,617.46	35.12
$K_G$	60,472.43	57,254.87	-3,217.55	-5.32
$\pi_{Di}$	22,485.03	27,432.55	4,947.52	22.00
$\pi_{Dn}$	74,096.70	153, 183.58	79,068.88	106.73
$E\left\{U^{D}\right\}$	559, 526.75	637, 876.55	78,349.80	14.00
$E\left\{U^{N}\right\}$	4, 166, 505.22	4, 115, 343.82	-51,161.40	- 1.23
$E\left\{S\right\}$	295,705.45	313,969.18	18,262.73	6.18

Table 2. Outcomes in the Baseline Setting, Except  $\alpha_T^{Dn} = -11.0$ .

Variable	No DG Charges	DG Charges	Change	% Change
r	196.85	196.73	-0.11	- 0.06
$w_i$	223.80	196.73	-27.07	- 12.10
w <sub>n</sub>	184.40	185.16	0.76	0.41
ki	0	6.62	N/A	N/A
$k_n$	0	-1.07	N/A	N/A
K <sub>Di</sub>	9,711.10	9,734.95	23.85	0.25
K <sub>Dn</sub>	6,170.57	6,107.17	-63.40	- 1.03
K <sub>G</sub>	27,350.57	27,374.23	23.66	0.09
$\pi_{Di}$	87,026.54	87, 454.55	428.00	0.49
$\pi_{Dn}$	112,719.09	110, 414.88	-2,304.20	-2.04
$E\left\{U^{D}\right\}$	725, 364.95	723,712.61	-1,652.34	- 0.23
$E\left\{U^{N}\right\}$	4,730,573.92	4,732,588.64	2,014.72	0.04
$E\left\{S\right\}$	539, 439.40	539,776.65	337.25	0.06

Table 3. Outcomes in the Coal Intensive Setting.

#### Appendix

Part A of this Appendix states and proves Proposition 6 and compares the findings in Propositions 3 and 6. Part B of this Appendix provides proofs of the propositions in the text.

#### A. TOU Prices and TOP DG Payments with Externalities.

Let [RP-*te*] denote the regulator's problem in the setting where the regulator sets TOU retail prices and TOP DG payments (and DG capacity charges) and where social losses from externalities ( $S_t(Q_t^v, Q_t^i, Q_t^n)$ ) are present. This problem is [RP- $\theta e$ ] with the additional restrictions specified in condition (8).

**Proposition 6.** At the solution to [RP-te], conclusion (i) in Proposition 3 holds. In addition, for  $t, t' \in \{L, H\}$   $(t' \neq t)$ :

$$\begin{aligned} (i) \quad \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ r_{jt} - \left( \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} + S_{tv}(\cdot) \right) \right] \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}} \, dF_{t}(\theta_{s}) \\ &+ \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ r_{jt'} - \left( \frac{\partial C_{t'}^{G}(Q_{t'}^{v}(\cdot,\theta_{s'}),K_{G})}{\partial Q_{t'}^{v}(\cdot,\theta_{s'})} + S_{t'v}(\cdot) \right) \right] \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}} \, dF_{t'}(\theta_{s'}) = 0 ; \\ (ii) \quad w_{nt} = \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} + S_{tv}(\cdot) - S_{tn}(\cdot) \right] dF_{t}(\theta_{s}); \quad and \\ (iii) \quad k_{y} = \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ \frac{\partial C_{t}^{G}(O_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} - w_{yt} + S_{tv}(\cdot) - S_{ty}(\cdot) \right] \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \, dF_{t}(\theta_{s}) - \frac{\partial T(\cdot)}{\partial K_{Dy}} \\ for \ y \in \{i,n\}. \end{aligned}$$

**Proof.** Let  $\lambda \ge 0$  denote the Lagrange multiplier associated with constraint (5). Then (1), (2), (3), (8), and the Envelope Theorem imply that that a solution to [RP-te], for  $y \in \{i, n\}$ ,  $j \in \{N, D\}$ , and  $t, t' \in \{L, H\}$   $(t' \ne t)$ :

$$K_{G}: \quad \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( -\frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}} \frac{dQ_{t}^{v}(\cdot)}{dK_{G}} - \frac{\partial C_{t}^{G}(\cdot)}{\partial K_{G}} \right) dF_{t}(\theta_{s}) - C^{K\prime}(K_{G}) - \frac{\partial T(\cdot)}{\partial K_{G}} \right] - \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} S_{tv}(\cdot) \frac{dQ_{t}^{v}(\cdot,\theta_{s})}{dK_{G}} dF_{t}(\theta_{s}) = 0; \quad (9)$$

$$k_y: \quad K_{Dy} - \sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left( S_{ty}(\cdot) \frac{\partial Q_t^y(\cdot,\theta_s)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_y} + S_{tv}(\cdot) \frac{dQ_t^v(\cdot,\theta_s)}{dk_y} \right) dF_t(\theta_s)$$

$$31$$

$$- \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( w_{yt} \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{d Q_{t}^{v}(\cdot)}{d k_{y}} \right) dF_{t}(\theta_{s}) + K_{Dy} + k_{y} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} \right] = 0; \quad (10)$$

$$w_{yt}: \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} Q_{t}^{y}(\cdot,\theta_{s}) dF_{t}(\theta_{s}) - \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( S_{ty}(\cdot) + S_{tv}(\cdot) \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \right) \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial w_{yt}} dF_{t}(\theta_{s}) - \sum_{t' \in \{L,H\}} \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left( S_{t'y}(\cdot) + S_{t'v}(\cdot) \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial Q_{t'}^{y}} \right) \frac{\partial Q_{t'}^{y}(\cdot,\theta_{s'})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}} dF_{t'}(\theta_{s'}) - \lambda \left[ \sum_{t' \in \{L,H\}} \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left( w_{yt'} \frac{\partial Q_{t'}^{y}(\cdot,\theta_{s'})}{\partial K_{Dy}} + \frac{\partial C_{t'}^{G}(\cdot)}{\partial Q_{t'}^{v}} \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial K_{Dy}} \right) \frac{\partial K_{Dy}}{\partial w_{yt}} dF_{t'}(\theta_{s'}) + \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( Q_{t}^{y}(\cdot,\theta_{s}) + w_{yt} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t'}^{v}} \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}} \right) dF_{t}(\theta_{s}) + k_{y} \frac{\partial K_{Dy}}{\partial w_{yt}} + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}} \right] = 0; \qquad (11)$$

$$R_{j}: -1 + \lambda = 0; \qquad (12)$$

$$\begin{split} r_{jt} &: \quad \int_{\underline{\theta}_{t}}^{\theta_{t}} \left( \left[ \frac{\partial V_{t}^{j}(X_{t}^{j}(\cdot,\theta_{s}),\theta_{s})}{\partial X_{t}^{j}} - r_{jt} \right] \frac{\partial X_{t}^{j}(\cdot)}{\partial r_{jt}} - X_{t}^{j}(\cdot) \right) dF_{t}(\theta_{s}) \\ &+ \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left( \frac{\partial V_{t'}^{j}(X_{t'}^{j}(\cdot,\theta_{s'}),\theta_{s'})}{\partial X_{t'}^{j}} - r_{jt'} \right) \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}} dF_{t'}(\theta_{s'}) \\ &- \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( S_{tv}(\cdot) \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial X_{t}} \frac{\partial X_{t}(\cdot,\theta_{s})}{\partial X_{t}^{j}} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}} \right) dF_{t}(\theta_{s}) \\ &- \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left( S_{t'v}(\cdot) \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial X_{t'}} \frac{\partial X_{t'}(\cdot,\theta_{s'})}{\partial X_{t'}^{j}} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}} \right) dF_{t'}(\theta_{s'}) \\ &+ \lambda \left[ \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( r_{jt} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}} + X_{t}^{j}(\cdot) \right) dF_{t}(\theta_{s}) + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} r_{jt'} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}} dF_{t'}(\theta_{s'}) \\ &- \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}), K_{G})}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial X_{t}^{j}} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}} dF_{t}(\theta_{s}) \end{split}$$

$$-\int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \frac{\partial C_{t'}^G(Q_{t'}^v(\cdot,\theta_{s'}),K_G)}{\partial Q_{t'}^v} \frac{\partial Q_{t'}^v(\cdot,\theta_{s'})}{\partial X_{t'}^j} \frac{\partial X_{t'}^j(\cdot,\theta_{s'})}{\partial r_{jt}} dF_{t'}(\theta_{s'}) \bigg] = 0.$$
(13)

(12) implies that  $\lambda = 1$ .  $\frac{\partial V_t^j(X_t^j(\cdot,\theta_s),\theta_s)}{\partial X_t^j} = r_{jt}$  for  $j \in \{D,N\}$  and  $t \in \{L,H\}$  since  $V_t^j(x,\theta_s)$  is the gross surplus consumer j derives from output x in state  $\theta_s$  in period t. Also,  $\frac{\partial X_t}{\partial X_t^j} = \frac{\partial Q_t^v}{\partial X_t^j} = \frac{\partial Q_t^v}{\partial X_t} = 1$  because  $Q_t^v = X_t^N + X_t^D - Q_t^i - Q_t^n$  and  $X_t = X_t^N + X_t^D$ . Therefore, (13) implies that conclusion (i) in the proposition holds.

Since  $\lambda = 1$  and  $\frac{dQ_t^v}{dK_G} = 0$ , (9) implies that conclusion (i) in Proposition 3 holds.

Observe that  $\lambda = 1$ ,  $\frac{dQ_t^v(\cdot,\theta_s)}{dK_{Di}} = -\mu_t \theta_s$ ,  $\frac{\partial K_{Di}}{\partial k_i}$  is not a function of  $\theta_s$ , and  $k_i$  only affects  $Q_t^v$  through  $K_{Di}$ . Therefore, for y = i, (10) can be written as conclusion (iii) in the proposition. The proof of conclusion (iii) for y = n is analogous.

Observe that  $\lambda = 1$ ,  $\frac{\partial K_{Dn}}{\partial w_{nt}}$  is not a function of  $\theta_{t'}$  for any  $t' \in \{L, H\}$ , and  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ . Therefore, for y = n, (11) can be written as:

$$-\left[\sum_{t'\in\{L,H\}}\int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}}\left(w_{nt'}-\frac{\partial C_{t'}^{G}(\cdot)}{\partial Q_{t'}^{v}}+S_{t'n}(\cdot)-S_{t'v}(\cdot)\right)\frac{\partial Q_{t'}^{n}(\cdot,\theta_{s'})}{\partial K_{Dn}}dF_{t'}(\theta_{s'})+k_{n}\right.\\ \left.+\frac{\partial T(\cdot)}{\partial K_{Dn}}\left[\frac{\partial K_{Dn}}{\partial w_{nt}}-\int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}}\left(w_{nt}-\frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}}+S_{tn}(\cdot)-S_{tv}(\cdot)\right)\frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial w_{nt}}dF_{t}(\theta_{s})\right]=0$$
$$\Rightarrow \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}}\left[w_{nt}-\frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}}+S_{tn}(\cdot)-S_{tv}(\cdot)\right]\frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial w_{nt}}dF_{t}(\theta_{s})\right]=0. \tag{14}$$

(14) reflects conclusion (iii) in the proposition for y = n. Since  $\frac{\partial Q_t^n(\cdot, \theta_s)}{\partial w_{nt}}$  is not a function of  $\theta_s$ , (14) implies that conclusion (ii) in the proposition holds.

A comparison of Propositions 3 and 6 reveals that the presence of social losses from externalities affects the optimal regulatory policy under TOU retail pricing and TOP DG compensation much as they do under state-specific retail pricing and DG compensation. In particular, retail prices are increased to discourage the purchase of electricity from the VIP as the social losses from externalities associated with electricity production by the VIP increase. In addition, DG payments for electricity produced using the non-intermittent technology increase as marginal social losses from externalities associated with this production decline relative to the marginal social losses associated with electricity production by the VIP, *ceteris paribus*. Furthermore, DG capacity charges are reduced as marginal losses from externalities associated with DG production decline relative to the corresponding losses associated with electricity production by the VIP.

#### B. Proofs of the Propositions in the Text.

**Proof of Proposition 1**. The proof follows immediately from the proof of Proposition 2. ■

#### Proof of Proposition 2.

Let  $\lambda \geq 0$  denote the Lagrange multiplier associated with constraint (5). Then (1), (2), (3), and the Envelope Theorem imply that at a solution to [RP- $\theta e$ ], for  $y \in \{i, n\}$  and  $j \in \{N, D\}$ :

$$k_{y}: \quad K_{Dy} - \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( S_{ty}(\cdot) \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} + S_{tv}(\cdot) \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} \right) dF_{t}(\theta_{s})$$
$$- \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( w_{yt}(\theta_{s}) \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{dQ_{t}^{v}(\cdot,\theta_{s})}{dk_{y}} \right) dF_{t}(\theta_{s})$$
$$+ K_{Dy} + k_{y} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} \right] = 0; \quad (15)$$

$$w_{yt}(\theta_{s}): \quad Q_{t}^{y}(\cdot,\theta_{s}) f_{t}(\theta_{s}) - \left[S_{ty}(\cdot) \left(\frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial w_{yt}(\theta_{s})} + \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})}\right) + S_{tv}(\cdot) \left(\frac{\partial Q_{t}^{v}(\cdot)}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}(\theta_{s})} + \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})}\right)\right] f_{t}(\theta_{s}) - \lambda \left[\left(w_{yt}(\theta_{s}) \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})} + Q_{t}^{y}(\cdot,\theta_{s}) + w_{yt}(\theta_{s}) \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial w_{yt}(\theta_{s})} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}(\theta_{s})}\right) f_{t}(\theta_{s}) + k_{y} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})} + \frac{\partial K_{Dy}}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})}\right] = 0; \quad (16)$$

34

$$r_{jt}(\theta_{s}): \left[ \left( \frac{\partial V_{t}^{j}(X_{t}^{j}(\cdot,\theta_{s}),\theta_{s})}{\partial X_{t}^{j}(\cdot,\theta_{s})} - r_{jt}(\theta_{s}) \right) \frac{\partial X_{t}^{j}(\cdot)}{r_{jt}(\theta_{s})} - X_{t}^{j}(\cdot) \right] f_{t}(\theta_{s}) \\ + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ \frac{\partial V_{t'}^{j}(X_{t'}^{j}(\cdot,\theta_{s'}),\theta_{s'})}{\partial X_{t}^{j}(\cdot,\theta_{s'})} - r_{jt'}(\theta_{s'}) \right] \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} dF_{t'}(\theta_{s'}) \\ - \left[ S_{tv}(\cdot) \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial X_{t}^{j}} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}(\theta_{s})} f_{t}(\theta_{s}) + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} S_{t'v}(\cdot) \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial X_{t'}^{j}} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} dF_{t'}(\theta_{s'}) \right] \\ + \lambda \left[ r_{jt}(\theta_{s}) \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}(\theta_{s})} + X_{t}^{j}(\cdot,\theta_{s}) - \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t'}^{v}} \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s})}{\partial X_{t'}^{j}(\cdot,\theta_{s})} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} \right] f_{t}(\theta_{s}) \\ + \lambda \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ r_{jt'}(\theta_{s'}) \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} - \frac{\partial C_{t'}^{G}(\cdot)}{\partial Q_{t'}^{v}(\cdot,\theta_{s'})} \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial X_{t'}^{j}(\cdot,\theta_{s'})} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} \right] dF_{t'}(\theta_{s'}) = 0. \quad (17)$$

(9) and (12) also hold. Therefore,  $\lambda = 1$  and conclusion (i) in Proposition 1 holds. Also,  $\frac{\partial Q_t^v}{\partial X_t^j} = 1$  and  $\frac{\partial V_t^j(X_t^j(\cdot,\theta_t),\theta_s)}{\partial X_t^j} = r_{jt}(\theta_s)$  for  $j \in \{D, N\}$ . Therefore, from (17), for each  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ :

$$\left[r_{jt}(\theta_{s}) - \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}(\cdot,\theta_{s})} - S_{tv}(\cdot)\right] \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}(\theta_{s})} f_{t}(\theta_{s}) + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[r_{jt'}(\theta_{s'}) - \frac{\partial C_{t'}^{G}(Q_{t'}^{v}(\cdot,\theta_{s'}),K_{G})}{\partial Q_{t'}^{v}(\cdot,\theta_{s'})} - S_{t'v}(\cdot)\right] \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} dF_{t'}(\theta_{s'}) = 0.$$
(18)

(18) is satisfied if:

$$r_{jt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot, \theta_s), K_G)}{\partial Q_t^v(\cdot, \theta_s)} + S_{tv}(\cdot) \text{ for all } \theta_s \in [\underline{\theta}_t, \overline{\theta}_t].$$
(19)

Observe that  $\lambda = 1$ ,  $\frac{\partial K_{Di}}{\partial k_{it}}$  is not a function of  $\theta_s$ , and  $k_i$  only affects  $Q_t^v$  through  $K_{Di}$ . Therefore, for y = i, (15) can be written as:

$$\left[\sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left(w_{it}(\theta_{s}) \mu_{t} \theta_{s} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{dQ_{t}^{v}(\cdot,\theta_{s})}{dK_{Di}} + S_{ti}(\cdot) \frac{\partial Q_{t}^{i}(\cdot,\theta_{s})}{\partial K_{Di}} + S_{tv}(\cdot) \frac{dQ_{t}^{v}(\cdot)}{dK_{Di}}\right) dF_{t}(\theta_{s}) + k_{i} + \frac{\partial T(\cdot)}{\partial K_{Di}} \left[\frac{\partial K_{Di}}{\partial k_{i}}\right] = 0.$$
(20)

Since  $\frac{dQ_t^v(\cdot,\theta_s)}{dK_{Di}} = -\mu_t \theta_s$  and  $\frac{\partial Q_t^i(\cdot,\theta_s)}{\partial K_{Di}} = \mu_t \theta_s$ , (20) can be written as:

$$k_{i} = -\frac{\partial T(\cdot)}{\partial K_{Di}} + \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}} - w_{it}(\theta_{s}) + S_{tv}(\cdot) - S_{ti}(\cdot) \right] \mu_{t} \theta_{s} dF_{t}(\theta_{s}).$$
(21)

Observe that  $\lambda = 1$ ,  $\frac{\partial K_{Dn}}{\partial k_{nt}}$  is not a function of  $\theta_s$ , and  $k_{nt}$  only affects  $Q_t^v$  through  $K_{Dn}$  and the corresponding impact on  $Q_t^n$ . Therefore, for y = n, (15) can be written as:

$$-\left[\sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left(w_{nt}(\theta_{s}) + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{n}} + S_{tn}(\cdot) + S_{tv}(\cdot) \frac{\partial Q_{t}^{v}(\cdot)}{\partial Q_{t}^{n}}\right) \frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial K_{Dn}} dF_{t}(\theta_{s}) + k_{n} + \frac{\partial T(\cdot)}{\partial K_{Dn}} \left[\frac{\partial K_{Dn}}{\partial k_{n}}\right] = 0.$$
(22)

Since  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ , (22) can be written as:

$$k_{n} = -\frac{\partial T(\cdot)}{\partial K_{Dn}} + \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left[ \frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}} - w_{nt}(\theta_{s}) + S_{tv}(\cdot) - S_{tn}(\cdot) \right] \\ \cdot \frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial K_{Dn}} dF_{t}(\theta_{s}).$$
(23)

Observe that  $\lambda = 1$ ,  $\frac{\partial Q_t^i(\cdot,\theta_s)}{\partial K_{Di}} = \mu_t \theta_s$ ,  $\frac{\partial Q_t^i(\cdot,\theta_s)}{\partial w_{it}(\theta_s)} = 0$ , and  $\frac{\partial Q_t^v(\cdot,\theta_s)}{\partial K_{Di}} = -\mu_t \theta_s$ . Therefore, for y = i, (16) can be written as:

$$\left[\left(w_{it}(\theta_s) - \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{ti}(\cdot) - S_{tv}(\cdot)\right)\mu_t \theta_s f_t(\theta_s) + k_i + \frac{\partial T(\cdot)}{\partial K_{Di}}\right] \frac{\partial K_{Di}}{\partial w_{it}(\theta_s)} = 0.$$

Therefore, for all  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ :

$$k_i = -\frac{\partial T(\cdot)}{\partial K_{Di}} + \left[\frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v} - w_{it}(\theta_s) + S_{tv}(\cdot) - S_{ti}(\cdot)\right] \mu_t \,\theta_s \, f_t(\theta_s) \,. \tag{24}$$

Observe that (21) and (24) are satisfied if:

$$w_{it}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v} + S_{tv}(\cdot) - S_{ti}(\cdot) \text{ for all } \theta_s \in [\underline{\theta}_t, \overline{\theta}_t].$$
(25)

(24) implies that if (25) holds, then:

$$k_i = -\frac{\partial T(\cdot)}{\partial K_{Di}}.$$
(26)

36

Recall that  $\lambda = 1$  and  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ . Therefore, for y = n, (16) can be written as:

$$\begin{bmatrix} w_{nt}(\theta_s) - \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{tn}(\cdot) - S_{tv}(\cdot) \end{bmatrix} \frac{\partial Q_t^n(\cdot, \theta_s)}{\partial K_{Dn}} \frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} f_t(\theta_s) + \begin{bmatrix} w_{nt}(\theta_s) - \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{tn}(\cdot) - S_{tv}(\cdot) \end{bmatrix} \frac{\partial Q_t^n(\cdot, \theta_s)}{\partial w_{nt}(\theta_s)} f_t(\theta_s) + k_n \frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} + \frac{\partial T(\cdot)}{\partial K_{Dn}} \frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} = 0.$$

Therefore, for each  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ :

$$\left[ \left( w_{nt}(\theta_s) - \frac{\partial C^G(\cdot)}{\partial Q^v} + S_{tn}(\cdot) - S_{tv}(\cdot) \right) \frac{\partial Q_t^n(\cdot, \theta_s)}{\partial K_{Dn}} f_t(\theta_s) + k_n + \frac{\partial T(\cdot)}{\partial K_{Dn}} \right] \frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} + \left[ w_{nt}(\theta_s) - \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{tn}(\cdot) - S_{tv}(\cdot) \right] \frac{\partial Q_t^n(\cdot, \theta_s)}{\partial w_n(\theta_s)} f_t(\theta_s) = 0.$$
(27)

Since  $\frac{\partial Q_t^n(\cdot,\theta_s)}{\partial K_{Dn}} > 0$ ,  $\frac{\partial Q_t^n(\cdot,\theta_s)}{\partial w_n(\theta_s)} > 0$ , and  $\frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} > 0$  for all  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ , (27) implies that if  $w_{nt}(\theta_s) > \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{tn}(\cdot) - S_{tv}(\cdot)$ , then  $k_n + \frac{\partial T(\cdot)}{\partial K_{Dn}} < 0$ . (27) also implies that if  $w_{nt}(\theta_s) < \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} + S_{tn}(\cdot) - S_{tv}(\cdot)$ , then  $k_n + \frac{\partial T(\cdot)}{\partial K_{Dn}} < 0$ . Therefore, because  $k_n + \frac{\partial T(\cdot)}{\partial K_{Dn}}$  does not vary with  $\theta_s$ , it must be the case that:

$$w_{nt}(\theta_s) = \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v} + S_{tv}(\cdot) - S_{tn}(\cdot) \text{ for all } \theta_s \in [\underline{\theta}_t, \overline{\theta}_t].$$
(28)

(23) and (28) imply:

$$k_n = -\frac{\partial T(\cdot)}{\partial K_{Dn}}. \quad \blacksquare \tag{29}$$

**Proof of Proposition 3**. The proof follows immediately from the proof of Proposition 6. ■

#### **Proof of Proposition 4**.

Let  $\lambda \geq 0$  denote the Lagrange multiplier associated with constraint (5). Then (1), (2), (3), and the Envelope Theorem imply that at a solution to [RP], for  $y \in \{i, n\}$ ,  $j \in \{N, D\}$ , and  $t \in \{L, H\}$ :

$$K_G: \quad \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left( -\frac{\partial C_t^G(\cdot)}{\partial Q_t^v} \frac{dQ_t^v(\cdot,\theta_s)}{dK_G} - \frac{\partial C_t^G(\cdot)}{\partial K_G} \right) dF_t(\theta_s) \right]$$

$$- C^{K'}(K_G) - \frac{\partial T(\cdot)}{\partial K_G} = 0; \qquad (30)$$

$$k_{y}: \quad K_{Dy} - \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( w_{y} \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{dQ_{t}^{v}(\cdot,\theta_{s})}{dk_{y}} \right) dF_{t}(\theta_{s}) + K_{Dy} + k_{y} \frac{\partial K_{Dy}}{\partial k_{y}} + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial k_{y}} \right] = 0; \quad (31)$$

$$w_{y}: \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\theta_{t}} Q_{t}^{y} dF_{t}(\theta_{s}) - \lambda \left[ \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( w_{y} \frac{\partial Q_{t}^{y}(\cdot,\theta_{s})}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{y}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{y}} \right. + Q_{t}^{y} + w_{y} \frac{\partial Q_{t}^{y}}{\partial w_{y}} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot)}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{y}} \right) dF_{t}(\theta_{s}) + k_{y} \frac{\partial K_{Dy}}{\partial w_{y}} + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{y}} \right] = 0; \qquad (32)$$

$$r: \sum_{j \in \{D,N\}} \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( \left[ \frac{\partial V_{t}^{j}(X_{t}^{j}(\cdot,\theta_{s}),\theta_{s})}{\partial X_{t}^{j}} - r \right] \frac{\partial X_{t}^{j}(\cdot)}{\partial r_{j}} - X_{t}^{j}(\cdot) \right) dF_{t}(\theta_{s})$$

$$+ \lambda \left[ \sum_{j \in \{D,N\}} \sum_{t \in \{L,H\}} \int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}} \left( r \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{j}} + X_{t}^{j}(\cdot) - \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial X_{t}^{j}} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r} \right) dF_{t}(\theta_{s}) \right] = 0. \quad (33)$$

(12) also holds, so  $\lambda = 1$ . Also,  $\frac{\partial Q_t^v}{\partial X_t^j} = 1$  and  $\frac{\partial V_t^j(X_t^j(r,\theta_s),\theta_s)}{\partial X_t^j} = r$  for  $j \in \{D, N\}$  and  $t \in \{L, H\}$ . Therefore, (33) implies that conclusion (i) of the proposition holds.

Since  $\lambda = 1$  and  $\frac{dQ_t^v}{dK_G} = 0$ , (30) implies that conclusion (i) in Proposition 3 holds.  $\lambda = 1$ ,  $\frac{\partial K_{Dn}}{\partial k_n}$  is not a function of  $\theta_s$ , and  $k_n$  only affects  $Q_t^v$  through  $K_{Dn}$  and the corresponding impact on  $Q_t^n$ . Therefore, for y = n, (31) can be written as:

$$-\left[\sum_{t \in \{L,H\}} \int_{\underline{\theta}_t}^{\overline{\theta}_t} \left(w_n + \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} \frac{\partial Q_t^v(\cdot,\theta_s)}{\partial Q_t^n}\right) \frac{\partial Q_t^n}{\partial K_{Dn}} dF_t(\theta_s) + k_n + \frac{\partial T(\cdot)}{\partial K_{Dn}}\right] \frac{\partial K_{Dn}}{\partial k_n} = 0.$$

Therefore, since  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ , conclusion (iii) in the proposition holds for y = n. The proof 38

of conclusion (iii) for y = i is analogous.

Recall that  $\lambda = 1$ ,  $\frac{\partial K_{Dn}}{\partial w_n}$  is not a function of  $\theta_t$ , and  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ . Therefore, for y = n, (32) can be written as:

$$-\left[\sum_{t\in\{L,H\}}\int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}}\left(w_{n}-\frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}}\right)\frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial K_{Dn}}dF_{t}(\theta_{s})+k_{n}+\frac{\partial T(\cdot)}{\partial K_{Dn}}\right]\frac{\partial K_{Dn}}{\partial w_{n}}$$
$$-\sum_{t\in\{L,H\}}\int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}}\left(w_{n}-\frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}}\right)\frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial w_{n}}dF_{t}(\theta_{s})=0$$
$$\Rightarrow\sum_{t\in\{L,H\}}\int_{\underline{\theta}_{t}}^{\overline{\theta}_{t}}\left[w_{n}-\frac{\partial C_{t}^{G}(Q_{t}^{v}(\cdot,\theta_{s}),K_{G})}{\partial Q_{t}^{v}}\right]\frac{\partial Q_{t}^{n}(\cdot,\theta_{s})}{\partial w_{n}}dF_{t}(\theta_{s})=0.$$
(34)

(34) reflects conclusion (iii) in the proposition for y = n. Since  $\frac{\partial Q_t^n}{\partial w_n}$  is not a function of  $\theta_s$ , (34) implies that conclusion (ii) of the proposition holds.

#### **Proof of Proposition 5**.

Let  $\lambda \geq 0$  denote the Lagrange multiplier associated with constraint (5). Then (1), (2), (3), and the Envelope Theorem imply that at a solution to [RP- $\theta k$ ], for  $j \in \{N, D\}$ ,  $t, t' \in \{L, H\}$   $(t' \neq t)$ , and  $y \in \{i, n\}$ :

$$r_{jt}(\theta_{s}): \left[ \left( \frac{\partial V_{t}^{j}(X_{t}^{j}(\cdot,\theta_{s}),\theta_{s})}{\partial X_{t}^{j}(\cdot,\theta_{s})} - r_{jt}(\theta_{s}) \right) \frac{\partial X_{t}^{j}}{r_{jt}(\theta_{s})} - X_{t}^{j}(\cdot) \right] f_{t}(\theta_{s}) + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ \frac{\partial V_{t'}^{j}(X_{t'}^{j}(\cdot,\theta_{s'}),\theta_{s'})}{\partial X_{t'}^{j}(\cdot,\theta_{s'})} - r_{jt'}(\theta_{s'}) \right] \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} dF_{t'}(\theta_{s'}) + \lambda \left[ r_{jt}(\theta_{s}) \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}(\theta_{s})} + X_{t}^{j}(\cdot,\theta_{s}) - \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot,\theta_{s})}{\partial X_{t}^{j}(\cdot,\theta_{s})} \frac{\partial X_{t}^{j}(\cdot,\theta_{s})}{\partial r_{jt}(\theta_{s})} \right] f_{t}(\theta_{s}) + \lambda \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ r_{jt'}(\theta_{s'}) \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} - \frac{\partial C_{t'}^{G}(\cdot)}{\partial Q_{t'}^{v}(\cdot,\theta_{s'})} \frac{\partial Q_{t'}^{v}(\cdot,\theta_{s'})}{\partial X_{t'}^{j}(\cdot,\theta_{s'})} \frac{\partial X_{t'}^{j}(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_{s})} \right] dF_{t'}(\theta_{s'}) = 0; \quad (35)$$

$$w_{yt}(\theta_s): \quad Q_t^y(\cdot, \theta_s) \ f_t(\theta_s) \\ - \ \lambda \left[ \left( w_{yt}(\theta_s) \frac{\partial Q_t^y(\cdot, \theta_s)}{\partial K_{Dn}} \frac{\partial K_{Dn}}{\partial w_{yt}(\theta_s)} + \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} \frac{\partial Q_t^v(\cdot, \theta_s)}{\partial Q_t^y} \frac{\partial Q_t^y(\cdot)}{\partial K_{Dn}} \frac{\partial K_{Dn}}{\partial w_{yt}(\theta_s)} \right] \right]$$

$$+ Q_{t}^{y}(\cdot,\theta_{s}) + w_{yt}(\theta_{s}) \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}(\theta_{s})} + \frac{\partial C_{t}^{G}(\cdot)}{\partial Q_{t}^{v}} \frac{\partial Q_{t}^{v}(\cdot)}{\partial Q_{t}^{y}} \frac{\partial Q_{t}^{y}(\cdot)}{\partial w_{yt}(\theta_{s})} \right) f_{t}(\theta_{s}) + \frac{\partial T(\cdot)}{\partial K_{Dy}} \frac{\partial K_{Dy}}{\partial w_{yt}(\theta_{s})} \bigg] = 0.$$
(36)

Since  $\lambda = 1$  and  $\frac{\partial Q_t^v}{\partial Q_t^n} = -1$ , (36) for y = n can be written as:

$$\left[ \left( w_{nt}(\theta_s) - \frac{\partial C^G(\cdot)}{\partial Q^v} \right) \frac{\partial Q_t^n(\cdot, \theta_s)}{\partial K_{Dn}} f_t(\theta_s) + \frac{\partial T(\cdot)}{\partial K_{Dn}} \right] \frac{\partial K_{Dn}}{\partial w_{nt}(\theta_s)} + \left[ w_{nt}(\theta_s) - \frac{\partial C_t^G(\cdot)}{\partial Q_t^v} \right] \frac{\partial Q_t^n(\cdot)}{\partial w_n(\theta_s)} f_t(\theta_s) = 0 \text{ for each } \theta_s \in [\underline{\theta}_t, \overline{\theta}_t].$$

Therefore, conclusion (ii) of the proposition for y = n holds for each  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]$ . The corresponding proof for y = i is analogous.

 $\frac{\partial V_t^j(X_t^j(r,\theta_s),\theta_s)}{\partial X_t^j} = r_{jt}(\theta_s) \text{ for each } j \in \{D,N\} \text{ and } t \in \{L,H\}. \text{ Also } \frac{\partial Q_t^v}{\partial X_t^j} = 1. \text{ Therefore,}$ since  $\lambda = 1$ , (35) implies that for each  $\theta_s \in [\underline{\theta}_t, \overline{\theta}_t]:$ 

$$\left[ r_{jt}(\theta_s) - \frac{\partial C_t^G(Q_t^v(\cdot,\theta_s), K_G)}{\partial Q_t^v(\cdot,\theta_s)} \right] \frac{\partial X_t^j(\cdot,\theta_s)}{\partial r_{jt}(\theta_s)} f_t(\theta_s) + \int_{\underline{\theta}_{t'}}^{\overline{\theta}_{t'}} \left[ r_{jt'}(\theta_{s'}) - \frac{\partial C_{t'}^G(Q_t^v(\cdot,\theta_{s'}), K_G)}{\partial Q_{t'}^v(\cdot,\theta_{s'})} \right] \frac{\partial X_{t'}^j(\cdot,\theta_{s'})}{\partial r_{jt}(\theta_s)} dF_{t'}(\theta_{s'}) = 0.$$
(37)

(37) is satisfied if conclusion (i) in the proposition holds.  $\blacksquare$ 

# References

American Council for an Energy-Efficient Economy, *Distributed Generation* (http://www.aceee.org/topics/distributed-generation), visited June 24, 2015.

Barbose, Galen, Samantha Weaver, and Naim Darghouth, *Tracking the Sun VII: An Historical Summary of the Installed Price of Photovoltaics in the United States from 1998 to 2013.* Lawrence Berkeley National Laboratory, September 2014.

Baumol, William and David Bradford, "Optimal Departures from Marginal Cost Pricing," *American Economic Review*, 60(3), June 1970, 265-283.

Borenstein, Severin, "The Private Net Benefits of Residential Solar PV: And Who Gets Them," Energy Institute at Haas, Working Paper 259, May 2015 (http://ei.haas.berkeley. edu/research/ papers/WP259.pdf).

Branker, K., M. Pathak, and J. Pearce, "A Review of Solar Photovoltaic Levelized Cost of Electricity," *Renewable and Sustainable Energy Reviews*, 15(9), December 2011, 4470-4482.

Brennan, Timothy, "Decoupling in Electric Utilities," *Journal of Regulatory Economics*, 38(1), August 2010a, 49-69.

Brennan, Timothy, "Optimal Energy Efficiency Policies and Regulatory Demand-Side Management Tests: How Well Do They Match?" *Energy Policy*, 38(8), August 2010b, 3874-3885.

Brown, David and David Sappington, "On the Design of Distributed Generation Policies: Are Common Net Metering Policies Optimal?" University of Alberta working paper, October 2015a (https://uofa.ualberta.ca/arts/about/people-collection/david-brown).

Brown, David and David Sappington, "On the Optimal Design of Demand Response Policies," University of Alberta working paper, November 2015b (https://uofa.ualberta.ca/arts/ about/people-collection/david-brown).

Brown, David and David Sappington, "Technical Appendix to Accompany 'Optimal Policies to Promote Efficient Distributed Generation of Electricity'," January 2016 (https://uofa.ual berta.ca/arts/about/people-collection/david-brown).

Brown, Toby and Ahmad Faruqui, "Structure of Electricity Distribution Network Tariffs: Recovery of Residual Costs," The Brattle Group Report Prepared for Australian Energy Market Commission, August 2014 (http://www.ksg.harvard.edu/hepg/Papers/2014/Brattle%20rep ort%20on% 20structure%20of%20DNSP%20tariffs%20and%20residual%20cost.pdf).

Bushnell, James, "Oligopoly Equilibria in Electricity Contract Markets," *Journal of Regulatory Economics*, 32(3), December 2007, 225-245.

California Energy Commission, *California Energy Almanac*, "Electricity Generation Capacity & Energy," (http://energyalmanac.ca.gov/electricity/electric\_generation\_capacity.html), visited September 10, 2015.

California Independent System Operator, *Open Access Same-time Information System*, 2015a, (http://oasis.caiso.com), visited September 15, 2015a.

California Independent System Operator, "Renewables Watch: Reports and Data," 2015b (http://www.caiso.com/market/Pages/ReportsBulletins/DailyRenewablesWatch.aspx).

California Public Utilities Commission, *Utility Tariff Information* (http://www.cpuc.ca.gov/PUC/energy/Electric+Rates/utiltariffs/index.htm), visited October 12, 2015.

California Solar Statistics, *Geographical Statistics* (https://www.californiasolarstatistics.ca. gov/reports/locale\_stats), visited September 10, 2015.

Chao, Hung-Po, "Demand Response in Wholesale Electricity Markets: The Choice of the Consumer Baseline," *Journal of Regulatory Economics*, 39(1), February 2011, 68-88.

Chu, Leon-Yang and David Sappington, "Motivating Energy Suppliers to Promote Energy Conservation," *Journal of Regulatory Economics*, 43(3), June 2013, 229-247.

Clean Power Research, *Minnesota Value of Solar: Methodology*, Prepared for Minnesota Department of Commerce, Division of Energy Resources, January 30, 2014 (https://www.clean power.com/wp-content/uploads/MN-VOS-Methodology-2014-01-30-FINAL.pdf).

Cohen, M., P. Kauzmann, and D. Callaway, "Economic Effects of Distributed PV Generation on California's Distribution System," Energy Institute at Haas Working Paper 260, June 2015 (http://ei.haas.berkeley.edu/research/papers/WP260.pdf).

Costello, Kenneth, "Major Challenges of Distributed Generation for State Utility Regulators," *The Electricity Journal*, 28(3), April 2015, 8-25.

Couture, Toby and Yves Gagnon, "An Analysis of Feed-in Tariff Remuneration Models: Implications for Renewable Energy Investment," *Energy Policy*, 38(3), February 2010, 955-965.

Darghouth, Naïm, Galen Barbose, and Ryan Wiser, "The Impact of Rate Design and Net Metering on the Bill Savings from Distributed PV for Residential Customers in California," *Energy Policy*, 39(9), September 2011, 5243-5253.

Darghouth, Naïm, Galen Barbose, and Ryan Wiser, "Customer-Economics of Residential Photovoltaic Systems (Part 1): The Impact of High Renewable Energy Penetrations on Electricity Bill Savings with Net Metering," *Energy Policy*, 67(4), April 2014, 290-300.

DNV GL Energy, A Review of Distributed Energy Resources. Prepared by DNV GL Energy for the New York Independent System Operator, September 2014 (www.nyiso.com/public/webdocs/media\_room/publications\_presentations/Other\_Reports/Other\_Reports/A\_Review\_of\_Distributed\_Energy\_Resources\_September\_2014.pdf).

Espey, James and Molly Espey, "Turning on the Lights: A Meta-Analysis of Residential Electricity Demand Elasticities," *Journal of Agricultural and Applied Economics*, 36(1), April 2004, 65-81.

Farrell, John, "Could Minnesota's "Value of Solar" Make Everyone a Winner?" *Renewable Energy World. Com*, March 22, 2014a (http://www.renewableenergyworld.com/rea/blog/post/2014/03/could-minnesotas-value-of-solar-make-everyone-a-winner).

Farrell, John, "Minnesota's Value of Solar: Can a Northern State's New Solar Policy Defuse Distributed Generation Battles?" Institute for Local Self-Reliance Report, April 2014b (http://www.ilsr.org/wp-content/uploads/2014/04/MN-Value-of-Solar-from-ILSR.pdf).

Faruqui, Ahmad and Ryan Hledik, "An Evaluation of SRP's Electric Rate Proposal for Residential Customers with Distributed Generation," The Brattle Group Report Prepared for Salt River Project, January 2015 (http://www.srpnet.com/prices/priceprocess/pdfx/DGRate Review.pdf).

Hledik, Ryan, "Rediscovering Residential Demand Charges," *The Electricity Journal*, 27(7), August - September 2014, 82-96.

ICF International, *Combined Heat and Power Policy Analysis and 2011 – 2030 Market Assessment*. Prepared for the California Energy Commission, February 2012 (http://www.energy.ca.gov/2012publications/CEC-200-2012-002/CEC-200-2012-002.pdf).

Kind, Peter, "Disruptive Challenges: Financial Implications and Strategic Responses to a Changing Retail Electric Business," Report prepared for the Edison Electric Institute, January 2013 (http://www.eei.org/ourissues/finance/Documents/disruptivechallenges.pdf).

Linvill, Carl, John Shenot, Jim Lazar, "Designing Distributed Generation Tariffs Well: Fair Compensation in a Time of Transition," The Regulatory Assistance Project, November 2013 (http://idahocleanenergy.org/wp-content/uploads/2013/01/2013-12-rap-carl-dg-tariffs-6898. pdf).

Lively, Mark and Lorraine Cifuentes, "Curing the Death Spiral: Seeking a Rate Design that Recovers Cost Fairly from Customers with Rooftop Solar," *Public Utilities Fortnightly*, August 2014.

London Economics International, "Estimating the Value of Lost Load," Briefing Paper Prepared for the Electric Reliability Council of Texas, June 17, 2013 (http://www.ercot.com/con tent/gridinfo/resource/2015/mktanalysis/ERCOT\_ValueofLostLoad\_LiteratureReviewand Macroeconomic.pdf).

Maine Public Utilities Commission (MPUC), "Central Maine Power Company Request for New Alternative Rate Plan: Revenue Allocation and Rate Design," Docket No. 2013 – 00168, 2013.

North Carolina Clean Energy Technology Center (NCCETC), *Personal Tax Credit by State*, 2015a (http://programs.dsireusa.org/system/program?type=31&technology=7&).

NCCETC, Solar Rebate Program by State, 2015b (http://programs.dsireusa.org/system/program?type=88&technology=7&).

NCCETC, Solar Renewable Energy Credit Program by State, 2015c (http://programs.dsireusa. org/system/program?type=85&).

NCCETC, The 50 States of Solar: A Quarterly Look at America's Fast-Evolving Distributed Solar Policy & Regulatory Conversation, 2015d (http://nccleantech.ncsu.edu/wpcontent/uploads/50-States-of-Solar-Issue2-Q2-2015-FINAL 3.pdf).

PacifiCorp, Integrated Resource Plan, Volume 1, April 30, 2013 (http://www.pacificorp.com/ content/dam/pacificorp/doc/Energy\_Sources/Integrated\_Resource\_Plan/2013IRP/Pacifi Corp-2013IRP\_Vol1-Main\_4-30-13.pdf).

Parsons and Brinckerhoff, *Electricity Transmission Costing Study*. An Independent Report Endorsed by the Institution of Engineering and Technology, 2012 (http://renewables-grid.eu/uploads/media/Electricity\_Transmission\_Costing\_Study\_Parsons\_Brinckerhoff.pdf).

Paul, Anthony, Erica Myers, and Karen Palmer, "A Partial Adjustment Model of U.S. Electricity Demand by Region, Season, and Sector," Resources for the Future Discussion Paper RFF DP 08-50, April 2009 (http://www.rff.org/files/sharepoint/WorkImages/Download/RF F-DP-08-50.pdf).

Perez-Arriaga, Ignacio and Ashwini Bharatkumar, "A Framework for Redesigning Distribution Network Use of System Charges Under High Penetration of Distributed Energy Resources: New Principles for New Problems," MIT Center for Energy and Environmental Policy Research Working Paper 2014-006, October 2014.

Poullikkas, Andreas, "A Comparative Assessment of Net Metering and Feed in Tariff Schemes for Residential PV Systems," *Sustainable Energy Technologies and Assessments*, 3, September 2013, 1-8.

Ramsey, Frank, "A Contribution to the Theory of Taxation," *Economic Journal*, 37(1), March 1927, 47-61.

Raskin, David, "The Regulatory Challenge of Distributed Generation," *Harvard Business Law Review Online*, 4, 2013, 38-51.

Schneider, Jordan and Rob Sargent, "Lighting the Way: The Top Ten States that Helped Drive America's Solar Energy Boom in 2013," PennEnvironment Research & Policy Center Report, August 2014 (http://environmentamericacenter.org/sites/environment/files/reports/ EA\_Lightingtheway\_scrn.pdf).

Shlatz, Eugene, Nathan Buch, and Melissa Chan, *Distributed Generation Integration Cost Study: Analytical Framework*. California Energy Commission Report, CEC-200-2013-007, Prepared by Navigant Consulting Inc., November 2013 (www.energy.ca.gov/2013publications /CEC-200-2013-007/CEC-200-2013-007.pdf).

Solar Energy Industries Association, *Solar State by State* (http://www.seia.org/policy/state-solar-policy, visited June 17, 2015).

Taylor, Thomas, Peter Schwarz, and James Cochell, "24/7 Hourly Response to Electricity Real-Time Pricing with up to Eight Summers of Experience," *Journal of Regulatory Economics*, 27(3), May 2005, 235-262.

U.S. Department of Energy, *Guide to Using Combined Heat and Power for Enhancing Reliability and Resiliency in Buildings*. Report Developed by the U.S. Department of Energy, the U.S. Environmental Protection Agency, and the U.S. Department of Housing and Urban Development, 2013 (https://www1.eere.energy.gov/manufacturing/distributedenergy/pdfs/chp\_critical\_facilities.pdf).

U.S. Energy Information Administration, *Reporting of Greenhouse Gases Program*, Table of Fuel and Energy Source: Codes and Emissions Coefficients. EIA Office of Integrated Analysis and Forecasting, 2014a (http://www.eia.gov/oiaf/1605/coefficients.html).

U.S. Energy Information Administration, *State Electricity Profiles*, Table 10. Supply and Disposition of Electricity, 2014b (http://www.eia.gov/electricity/state/).

U.S. Energy Information Administration, "Levelized Cost and Levelized Avoided Cost of New Generation Resources," *Annual Energy Outlook*, 2015a (http://www.eia.gov/forecasts/ aeo/pdf/electricity\_generation.pdf).

U.S. Energy Information Administration, *Form 826*, 2015b (http://www.eia.gov/electricity/ data/eia826).

U.S. Energy Information Administration, *Detailed State Data*, 2015c (http://www.eia.gov/elec tricity/data/state).

U.S. Environmental Protection Agency (EPA), Regulatory Impact Analysis for the Proposed Standards of Performance for Greenhouse Gas Emissions for New Stationary Sources: Electric Utility Generating Units, EPA-452/R-13-003, September 2013 (http://www2.epa.gov/sit es/production/files/2013-09/documents/20130920proposalria.pdf).

Virginia State Corporation Commission (VSCC), "Application of Virginia Electric and Power Company For Approval of a Standby Charge and Methodology and Revisions to its Tariff and Terms and Conditions of Service," Case No. PUE-2011-00088, 2011 (http://www.scc.virginia.gov/docketsearch/DOCS/2hm201!.PDF).

Wade, Steven, "Price Responsiveness in the AEO2003 NEMS Residential and Commercial Buildings Sector Models," Report Prepared for the U.S. Energy Information Administration, 2003 (http://www.eia.gov/oiaf/analysispaper/elasticity/pdf/buildings.pdf).

Weissman, Steven and Nathaniel Johnson, *The Statewide Benefits of Net-Metering in California & the Consequences of Changes to the Program*, Center for Law, Energy & the Environment, University of California, 2012 (http://www.law.berkeley.edu/files/The\_Statewide\_ Benefits\_of\_Net-Metering\_in\_CA\_Weissman\_and\_Johnson.pdf). Wiser, Ryan, Andrew Mills, Galen Barbose, and William Golove, "The Impact of Retail Rate Structures on the Economics of Commercial Photovoltaic Systems in California," Lawrence Berkeley National Laboratory LBNL – 63019 Report, July 2007. http://emp.lbl.gov/sites/all/files/ REPORT%20lbnl%20-%2063019\_0.pdf.

World Alliance for Decentralized Energy, *Where Can DE be Used?* (http://www.localpower. org/deb\_where.html), visited September 21, 2014.

Yamamoto, Yoshihiro, "Pricing Electricity from Residential Photovoltaic Systems: A Comparison of Feed-in Tariffs, Net Metering, and Net Purchase and Sale," *Solar Energy*, 86(9), September 2012, 2678-2685.

# Department of Economics, University of Alberta Working Paper Series

**2015-18:** Departure and Promotion of U.S. Patent Examiners: Do Patent Characteristics Matter? - Langinier, C., Lluis, S.

**2015-17:** Socioeconomic Inequalities in Infant Mortality in Egypt: Analyzing Trends between 1995 and 2014 – **Sharaf, M.**, Rashad, A.

**2015-16:** Does Economic Growth Reduce Child Malnutrition in Egypt? New Evidence from National Demographic and Health Survey – Rashad, A., **Sharaf, M.** 

**2015-15:** The Labor Market and School Finance Effects of the Texas Shale Boom on Teacher Quality and Student Achievement – **Marchand**, **J.**, Weber, J.

**2015-14:** Measuring Market Power and the Efficiency of Alberta's Restructured Electricity Market: An Energy-Only Market Design – **Brown**, **D.**, Olmstead, D.

**2015-13:** The Welfare and Stabilization Benefits of Fiscal Rules: Evidence from Canadian Provinces - Landon, S., Smith, C.

**2015-12:** Law and Economics and Tort Litigation Institutions: Theory and Experiments - Landeo, C.

2015-11: Effective Labor Relations Laws and Social Welfare - Landeo, C., Nikitin, M.

**2015-10:** Stipulated Damages as a Rent-Extraction Mechanism: Experimental Evidence - Landeo, C., Spier, K.

2015-09: Incentive Contracts for Teams: Experimental Evidence - Landeo, C., Spier, K.

2015-08: Financially-Constrained Lawyers - Landeo, C., Nikitin, M.

**2015-07:** Plant Breeders' Rights, Patents and Incentives to Innovate - Hervouet, A., Langinier, C.

**2015-06:** Physical Activity, Present Bias, and Habit Formation: Theory and Evidence from Longitudinal Data - Humphreys, B., Ruseski, J., **Zhou, L.** 

**2015-05:** The Impact of Female Education on Teenage Fertility: Evidence from Turkey – Günes, P.

2015-04: Credit Constraints, Technology Upgrading, and the Environment – Andersen, D.

2015-03: On the Optimal Design of Demand Response Policies – Brown, D., Sappington, D.

**2015-02:** Fiscal Transparency, Measurement and Determinants: Evidence from 27 Developing Countries – Tehou, Y., **Sharaf, M.** 

**2015-01:** Trends in Earnings Inequality and Earnings Instability among U.S. Couples: How Important is Assortative Matching? – **Hryshko**, **D.**, Juhn, C., McCue, K.

**2014-13:** Capacity Payment Mechanisms and Investment Incentives in Restructured Electricity Markets – **Brown**, **D**.

**2014-12:** On the Optimal Design of Distributed Generation Policies: Is Net Metering Ever Optimal? – **Brown, D.,** Sappington, D.

**2014-11:** Output Growth and Commodity Prices in Latin America: What Has Changed? – **Fossati**, **S**.

**2014-10:** A Survey of the Economics of Patent Systems and Procedures – **Eckert**, **A**., **Langinier**, **C**.

**2014-09:** Using Bayesian Imputation to Assess Racial and Ethnic Disparities in Pediatric Performance Measures – **Brown, D.,** Knapp, C., Baker, K., Kaufmann, M.

**2014-08:** Effects of an Integrated Care System on Children with Special Heath Care Needs' Medicaid Expenditures – Marcu, M., Knapp, C., Madden, V., **Brown, D.,** Wang, H., Sloyer, P.

**2014-07:** The Effect of Subsidized Entry on Capacity Auctions and the Long-Run Resource Adequacy of Electricity Markets – **Brown**, **D**.